# Inertial Sensors for Human Motion Analysis: A Comprehensive Review

Sara García-de-Villa<sup>(D)</sup>, David Casillas-Pérez<sup>(D)</sup>, Ana Jiménez-Martín<sup>(D)</sup>, *Member, IEEE*, and Juan Jesús García-Domínguez<sup>(D)</sup>, *Senior Member, IEEE* 

Abstract—Inertial motion analysis is having a growing interest during the last decades due to its advantages over classical optical systems. The technological solution based on inertial measurement units allows the measurement of movements in daily living environments, such as in everyday life, which is key for a realistic assessment and understanding of movements. This is why research in this field is still developing and different approaches are proposed. This presents a systematic review of the different proposals for inertial motion analysis found in the literature. The search strategy has been carried out on eight different platforms, including journal articles and conference proceedings, which are written in English and published until August 2022. The results are analyzed in terms of the publishers, the sensors used, the applications, the monitored units, the algorithms of use, the participants of the studies, and the validation systems employed. In addition, we delve deeply into the machine learning techniques proposed in recent years and in the approaches to reduce the estimation error. In this way, we show an overview of the research carried out in this field, going into more detail in recent years, and providing some research directions for future work.

*Index Terms*—Human motion, inertial measurement units (IMUs), inertial sensors, kinematic analysis, motion analysis.

## I. INTRODUCTION

**H**UMAN motion analysis is an essential support tool for the assessment of the parameters of movements, which is especially important in the evaluation of workout routines, clinical rehabilitation, and preventive treatments [1]. It is also

Manuscript received 2 November 2022; revised 12 April 2023; accepted 30 April 2023. Date of publication 15 May 2023; date of current version 5 June 2023. This work was supported in part by the Junta de Comunidades de Castilla La Mancha through the FrailAlert project under Grant SBPLY/21/180501/000216, in part by the Spanish Ministry of Science and Innovation through the INDRI project under Grant PID2021-122642OB-C41, in part by the Comunidad de Madrid through the RACC project under Grant CM/JIN/2021-016, and in part by the NEXTPERCEPTION European Union project funded by ECSEL Joint Undertaking (JU) under Grant Agreement 876487 (ECSEL-2019-2-RIA). The Associate Editor coordinating the review process was Bruno Andò. (*Corresponding author: Sara García-de-Villa.*)

Sara García-de-Villa was with the Department of Electronics, University of Alcalá, 28805 Alcalá de Henares, Spain, and also with the Center of Automation and Robotics (CSIC-UPM), 28500 Arganda del Rey, Spain. She is now with the Department of Signal Theory and Communications, Rey Juan Carlos University, 28942 Madrid, Spain (e-mail: sara.garcia.devilla@urjc.es).

David Casillas-Pérez is with the Department of Signal Theory and Communications, Rey Juan Carlos University, 28942 Madrid, Spain (e-mail: david.casillas@urjc.es).

Ana Jiménez-Martín and Juan Jesús García-Domínguez are with the Department of Electronics, University of Alcalá, 28805 Alcalá de Henares, Spain (e-mail: ana.jimenez@uah.es; jjesus.garcia@uah.es).

Digital Object Identifier 10.1109/TIM.2023.3276528

becoming very popular for physical activity monitoring in the elderly. Indeed, as the population of developed countries ages, the demand for home-based rehabilitation and the need to obtain quantitative exercise data remotely will increase [2].

Optical methods are considered the *gold standard* in the motion analysis field because of their accurate measurements of kinematic and spatiotemporal parameters [3]. However, these systems entail several disadvantages, such as the high cost of equipment, the need for trained personnel to use the equipment, the required large spaces for installation, and their restricted margin of maneuverability, which limits their use in controlled indoor environments.

The inertial motion analysis has emerged as a promising alternative to optical methods attracting great scientific interest. Inertial systems are portable and can be used everywhere, which means an advantage to the optical systems, which are commonly constraint to a limited space. That makes the inertial measurement units (IMUs) an affordable and friendly use alternative for the estimation of human kinematics. These devices allow continuous monitoring of human motions in daily environments, which is crucial in order to obtain more reliable information than the obtained in sporadic laboratory tests. For these reasons, the use of IMUs has increased in the last few decades for continuous monitoring of human motions, as reported in [4].

Previous works extensively review the use of portable sensors. A recent review analyzes the integration of portable sensors in clothes to obtain physiological and motion information [5]. However, inertial sensors are not considered in the analysis, in spite of their frequent use in this field. Reviews that take into account the use of inertial sensors are focused on applications, such as sign languages or motion analysis [3], [6]. Works about the inertial motion analysis, as in [3], address the motion monitoring and kinematic feature extraction, but only considering the specific area of sportrelated exercises evaluation and their analysis is up to April 2017. A lower limb-focused study is carried out in [7], but it does not provide a complete overview of the literature on inertial motion analysis. To the best of our knowledge, the last in-depth and generic systematic review on inertial sensors for human motion analysis is reported in [4], published in 2016. Since the number of publications about human motion analysis increases over time, as shown in Fig. 1, we consider that there is a need to update the literature review on this topic. According to Fig. 1, the number of existing publications on

This work is licensed under a Creative Commons Attribution 4.0 License. For more information, see https://creativecommons.org/licenses/by/4.0/



Fig. 1. Number of publications focused on the inertial motion analysis, referred to obtaining kinematic parameters by using portable inertial sensors, found in the literature.

the inertial motion analysis field has considerably increased since the previous review was published [4].

Furthermore, during the last years, machine learning (ML) methods have arisen, and they have been applied to inertial motion analysis. Consequently, it is required an update to provide an overview of the algorithms analyzed in [4], such as the Kalman filters (KFs), complementary filters (CFs), integration, and vector observation, but in combination with the novel ML-based approaches.

For these reasons, the main aim of this work is to review the current state of inertial sensors for human monitoring, especially considering the occurrence and evolution of ML methods for this research field. Another objective of this work is to analyze the current trends and provide insights into inertial motion analysis. To do so, we review the published works on human motion analysis using IMUs and analyze the selected ones in terms of: 1) publisher and years; 2) sensors used; 3) type of estimations referred to the dimensions of the estimated magnitudes; 4) the aimed applications of the proposals; 5) the monitored motion units; 6) the algorithmic approaches, with an in-depth analysis of sensor fusion filters, data science algorithms, and the approaches for error reduction; 7) the study participants; and 8) the validation systems (VSs) and metrics. Finally, on the basis of the findings, we suggest future research directions.

The rest of this document is structured as follows. Section II describes the search strategy and the eligibility criteria applied in this work; Section III details and analyzes the findings according to the terms explained earlier. Section IV discusses the general trends of the studied works and analyzes the future directions. Finally, Section V summarizes the main contributions of this work.

## II. MATERIALS AND METHODS

In this section, we describe the workflow to search and select the works in the state of the art included in this review. We describe this article's screening process and analyze the common publishers of this field. Finally, we detail the data extracted from them for further analysis.

# A. Eligibility Criteria

This review focuses on peer-reviewed articles, book chapters, and conference papers. Papers are required to be published in English and describe the methodology employed

TABLE I DATABASES CONSULTED IN THE LITERATURE SEARCH

Database	Source	# papers
ACM Digital Library	dl.acm.org	17
IEEE Xplore	ieeexplore.ieee.org	145
PubMed	www.ncbi.nlm.nih.gov/pubmed	188
ScienceDirect	www.sciencedirect.com	115
Scopus	www.scopus.com/	1434
Taylor & Francis Online	www.tandfonline.com	2
Web of Science	webofknowledge.com	326
Wiley Online Library	onlinelibrary.wiley.com	21

to obtain human kinematic parameters using only IMUs. Sensor fusion with other devices is not considered. This review only includes those papers that validate their results using a reference system. If a journal article is an extended version of a conference one, only the journal article is included.

## B. Literature Search Strategy

Considering the eligibility criteria, we select eight databases (ACM Digital Library, IEEE Xplore, PubMed, Science Direct, Scopus, Taylor & Francis Online, Web of Science, and Wiley Online Library) for the search of related papers (see Table I). Following the strategy of the previous review [4] about this topic, we use the same search command, which consists of:

# ("human motion" OR "human movement") AND ("wearable sensors" OR "inertial sensors" OR "wearable system")

considering their presence in the title, abstract, or keywords. The search includes journals, book chapters, or conference proceedings. This article abstract is required to be available during this search. No restriction was imposed on the date of publication.

The initial search on the databases in Table I leads to a review of 2 248 papers. The papers found in this search do not include important references from the state of the art, such as [8] or [9], so we expand the search. The new search is only performed on the Scopus website since it is the largest database of all those evaluated (see Table I). In this case, we use the following command, which is less restrictive than the previous one:

[("human motion" OR "human movement" OR "joint kinematics" OR "body tracking" OR kinematic\* OR "joint angle\*" OR "joint angle velocity" OR "joint angle acceleration") AND (IMU OR "inertial sensors" OR "inertial measurement unit" OR accelerometer OR gyroscope OR magnetometer)].

The search criteria are to find these key phrases in the title, abstract, or keywords of articles. This second search adds 1882 documents, so we finally obtain 4130 papers to review.

Starting from the results of this search, we carry out a Preferred Reporting Items for Systematic reviews and Meta-Analyses (PRISMA) screening process [10] to determine the documents included in this study. Fig. 2 depicts the processes of identification and screening to determine the works included in this review.

After excluding duplicated citations, the number of documents to screen is reduced to 3775 papers. The results of



Fig. 2. PRISMA search strategy flowchart.

this search include works in the field of human motion analysis with IMUs. However, this IMU-based motion analysis covers a wide range of topics, such as the estimation of kinematic and spatiotemporal parameters or the motion-based evaluation of health, as depicted in Fig. 3.

The topics of kinematic and spatiotemporal parameters refer to the analysis of different motion magnitudes, such as joint angles, trajectory, or speed, whereas human body calibration includes the location of joints or the estimation of segment lengths. The last two topics, human monitoring and motionbased evaluation, are focused on qualitative analysis, such as recognizing types of motions or activities and identifying behavior patterns. Our study is focused on wearable inertial sensors and kinematic parameters as joint rotation angles, so we discard by abstract reading those works that are focused on any other topic. In this way, we exclude 3262 papers for not being related to the topic of this review.

We found 513 potentially relevant studies of this topic for quality assessment. To consider a study in this review, we set the inclusion criteria detailed in Fig. 2, which are referred to the proposed or applied algorithm, its validation, and the sensor system used. Finally, 147 studies meet the inclusion criteria and are analyzed in this review.

#### C. Publisher and Years

Most reviewed works are journal articles (72.1%) [see Fig. 4 (top)]. These works are published in 42 journals. 56.7% of them appear in sevenjournals (each of them with at least four papers), as shown in Fig. 4 (bottom). The journals that appear with the highest frequency in this search are *Sensors*, IEEE SENSORS JOURNAL (IEEE SJ), IEEE TRANSACTIONS ON BIOMEDICAL ENGINEERING (IEEE TBE), *Journal of Biomechanics* (JBiomech), *Gait & Posture* (G&P), IEEE TRANSACTIONS ON INSTRUMENTATION AND MEASUREMENT (IEEE TIM), and IEEE JOURNAL OF

BIOMEDICAL AND HEALTH INFORMATICS (IEEE JBHI). The remaining 43.3% of works are distributed in 35 journals.

There is a clear increasing interest in the research field of inertial motion capture (see Fig. 1). This review is not restricted to any date in order to analyze all the works related to this topic and provide a general overview and its evolution. Fig. 1 shows the number of papers published during the fiveyear periods from 1991 to August 2022. Only twoworks are dated on the first studied decade, 21 works on the second one, and 98 works on the third one. There are also 27 works published in the period 2021–2022, the last years studied in this review.

Since 2016, the last year studied in [4], the number of publications has highly (see Fig. 5). These figures support the need for an update of a systematic review on the research topic of human motion analysis by using IMUs.

#### D. Data Extraction, Analysis, and Examination

We categorize the selected papers in terms of a set of relevant details. We classify them into two groups in order to ease the study and its reading. First, we evaluate the details related to the implemented algorithms, the sensors in use, and the estimations. Second, we study the specific anatomic part of the human body studied in each work, the VS and metrics used, and the information related to the validation subjects.

Regarding the first set of details, we analyze the following parameters: the fusion algorithm (FA) implemented for the motion analysis, which indicates with "SF" if the work uses sensor fusion approaches, "ML" the application of ML techniques, and "OA" any other proposal; the use of biomechanical constraints (BCs) and their related requirements of anatomical information (ANT) as the segment lengths or the joint location with respect to the IMU sensors; the implementation of other corrections (OCs); the type of sensor used to measure the motions (GS: gyroscope sensor, AS: accelerometer sensor, and MS: magnetometer sensor) and the use of external sensors to train ML-based algorithms, but not in the motion prediction (OS); the type of estimation (EST), considering the possible planar (2-D) or 3-D estimations; and the measured magnitude (ANG: angle or DIS: displacement referred to the change in the position of the corresponding point, i.e., the sensor or the monitoring joint) and the monitored motion unit (JNT: joint or SGM: segment). These details are shown in Table III in Appendix A for the selected papers.

With respect to the human body part, we study the lower group (LG) or upper group (UG) of segments and joints. We also report the VS used as *ground truth* in the studied works and the metrics (RMSE: root mean square error; nRMSE: normalized RMSE; %RMSE: percentage of RMSE; MAE: mean absolute error; AE: average error; CFC: correlation coefficient; LAMs: limits of agreement; MV: maximum variation; accuracy; and error rate), labeled as **M1** and **M2** in Table IV in Appendix A, employed in the proposed methods. Finally, we provide the number of subjects (NS) studied if this population considered presents a motor-related disease (**DSS**). Table IV in Appendix A includes the details of these parameters explained earlier in the selected papers.



Human motion analysis with Magneto-Inertial Measurement Units





Fig. 4. Distribution of the papers with respect to the type of publication environment in which they were published. Top: conference and journal distribution. Bottom: journals that published the analyzed works.

#### **III. REVIEW FINDINGS**

Based on the categorization of the papers with respect to their relevant characteristics, as presented in Tables III and IV in Appendix A, in this section, we describe the main findings.

# A. Sensors

IMUs contain triaxial gyroscopes, accelerometers, and, commonly, magnetometers. The information from these sensors is used separately through the observation of vectors, as gravity in the accelerometer data or the magnetic field in magnetometers, or by integration of the gyroscope data. Another approach is to gather their measurements in different combinations of two or three sensors with different algorithms, such as sensor fusion filters or ML methods. In order to



Fig. 5. Year of publication of the reviewed papers. Top: trend from the first motion analysis-related work until nowadays. Bottom: distribution of publications in the last five-year period.



Fig. 6. Sensor type and combination used in the analyzed works (AS: accelerometer sensor; GS: gyroscope sensor; and MS: magnetometer sensor).

illustrate the proportion of their utilization, separately or fused, Fig. 6 shows the percentage of use of each sensor or combination.

The integration of the turn rate alone entails inherent errors. In the estimations of kinematic parameters, the turn rate integration results in an accumulated error from the gyroscope bias. For that reason, only 4.8% of studies use this sensor alone [11], [12], [13], [14], [15], [16], [17].

Accelerometers are more frequently used separately (8.8%). Their measurement of specific force allows us to obtain a direct observation of the gravity vector, used as orientation reference [18], [19], [20], [21], [22], [23], [24], [25], [26], [27], [28], [29], [30]. However, the direct observation of the gravity vector is only possible when accelerometers are static, as in gait strides during the stance phase.

Magnetometers are the most limited sensor analyzed because of their sensitivity to magnetic disturbances in the environment. As a consequence, only 1.4% of studies use this sensor independently [31], [32].

Sensor fusion techniques are useful methods to overcome the individual limitations of each of them separately. Most of studies that fuse data from various sensors combine gyroscopes and accelerometers [8], [33], [34], [35], [36], [37], [38], [39], [40], [41], [42], [43], [44], [45], [46],[47], [48], [49], [50], [51], [52], [53], [54], [55], [56], [57], [58], [59], [60], [61], [62], [63], [64], [65], [66], [67], [68], [69], [70], [71], [72], [73], [74], [75], [76], [77], [78], [79],[80], [81], [82], [83], [84], [85], [86], [87], [88], [89], [90], [91], [92], [93], [94], [95], [96], [97], [98], [99], [100], [101], [102], [103], [104], [105], [106], [107], [108], [109],[110], [111], [112], [113], [114], [115] or both sensors with magnetometers [116], [117], [118], [119], [120], [121], [122], [123], [124], [125], [126], [127], [128], [129], [130], [131], [132], [133], [134], [135], [136], [137], [138], [139], [140], [141], [142], [143], [144], [145], [146], [147], [148], [149], [150], [151], [152]. Few studies join the accelerometer and magnetometer data [153], [154], and only one uses the gyroscope and magnetometer data [155].

The 3-D position of devices can be inferred from the IMU sensor information. The combination of the three sensors, gyroscope, magnetometer, and accelerometer includes information on the angular rate of motion and the references of the vector gravity and Earth's magnetic field references. However, these 3-D positions can also be estimated by sensor fusion techniques using different combinations of the three sensors in IMUs. Most studies give 3-D estimations (71.4%), as shown in Fig. 7. Conversely, only 1.4% of works use accelerometers and magnetometers, and 25.2% of works use both sensors complemented with gyroscopes (see Fig. 6). This fact is noticeable because the combination of the first two sensors is required to obtain the references to overcome the gyroscope drift and get accurate 3-D estimations. It implies that the majority of algorithms that offer 3-D space predictions propose methods for error reduction, which do not rely on vector references. In this way, the magnetic disturbances that cause errors in the magnetic field measurements are avoided.

The 2-D estimations include kinematic parameters in any plane perpendicular to the floor and the two angles with respect to the horizontal plane, in the frontal and sagittal planes. No doubt that the 3-D estimations are more complete since they provide information about the whole motion, even if it is mostly performed in one plane. That is the reason why only 27.9% of studies focus on obtaining estimations in the 2-D space.



Fig. 7. Type of estimations in terms of their dimensionality, divided between 2-D and 3-D estimations.

Only one study adapts the estimation to 3-D or 2-D spaces according to the motions [41]. In this proposal, the method gives 2-D estimations based on the accelerometer data when the motion is mostly performed in one plane. If deviations from this plane are detected, the method integrates the gyroscope data in order to provide 3-D estimations.

#### **B.** Application

Healthcare applications are the most common ones (95.2%) in the inertial motion analysis field (see Fig. 8). These applications include motion capture or analysis, gait and clinical assessment, or rehabilitation. The aim of 33.5% of studies is to motion capture to obtain information about human kinematics for motion analysis or find possible diseases. Gait is the second most common application (19.1%) due to its relationship with cognitive impairments. The prevalence of use for the specific clinical assessment is similar, being the aim of the 14.5% of works. Rehabilitation and sports are also worth mentioning because they are very motivating in research works (19.6% of studies).

#### C. Monitored Motion Unit

We analyze the anatomical unit measured in the reviewed works. In this work, the anatomical units are called monitored motion units following the nomenclature of previous studies [4]. We divide these monitored motion units into two groups: segments and joints. Segments usually correspond to elements of the skeletal system, such as thighs (femur), and are modeled as rigid-solid bodies. Joints are the unions between segments. The objective of 64.6% of studies is to measure the motion of joints, whereas 27.2% of proposals focus on tracking segments, and the remaining 8.2% combine the monitoring of both monitored motions, segments, and joints, as shown in Fig. 9 (top).

Studies focus more frequently on the lower half (61.2%) of the body than on than upper half (34.7%). Compared to the outcomes in the review of Lopez-Nava et al. [4], this trend is in the most recent works different than in the previous ones. We found that recent research, dated the last three years, extends motion analysis to full-body monitoring, which is an important difference from the findings of previous studies [4]. We consider full body if both upper and lower halves are monitored, which is made in the 4.1% of studies. Fig. 10 (left) depicts the percentage of works that monitor each body half or the full body.



Fig. 8. Analysis of the application of the studied works. Left: percentage of proposals whose aimed field is included in healthcare-related applications. Right: specific application or applications considered in proposals. Some works provide possible uses of their proposals, others focus on a specific application, such as gait, and others refer to the general motion capture field. FES refers to functional electrical stimulation, and the research of human–robot interactions (HRIs) is labeled as "human–robot."



Fig. 9. Monitored motion units and the obtained measurement. Top: percentage of studies that measure each motion unit or their combination. Bottom: type of measurement, orientation, and location.

We study the groups of segments or joints included in each body half for a deeper analysis. We define the groups as sets of monitoring units. We consider that studies focus on one of the groups if they estimate the orientation or location of one of the included monitored units. For example, one study that tracks the motion of wrists is included in the hand group since we establish that the hand group includes the wrist among other motion units.

With respect to the upper half of the body, we divide it into the hand (hand, wrist, and fingers), arm segments (arm and forearm), arm joints (shoulder, elbow, and forearm twist), trunk (back, trunk, and torso), and head and upper back (head/neck/scapula). The upper half groups (51 works) are named follows: arm segments (U1), trunk (U2), arm joints and hand (U3), head and upper back, trunk, arm segments and arm joints (U4), arm joints (U5), head and upper back (U6), arm segments and joints (U7), trunk, arm joints and hand (U8), head and upper arm, trunk and arm (U9), head and upper arm and arm joints (U10), head and upper back and trunk (U11), head and upper back and arm segments (U12), and head and upper back alone (U13). Fig. 10 (right) shows the presence of the combination of these groups in the studied works.

The arm joints group, U5, is the one on which most works are focused (12/51 studies), followed by the trunk (9/51 studies), U2. The next three frequent groups are the arm segments (7/51 studies), U1, the combination of arm and hand joints (7/51 studies), U3, and the head and upper back (6/51 studies), U6. The rest of the groups are only monitored in 1/51 or 2/51 studies according to the case. In this way, research works commonly focus on the study of the arms more than the other upper half body structures.

With regard to the lower half, we divide it into the pelvis, leg segments (thigh and shin), leg joints (hip/knee/ankle), and feet. The names of the groups of their combinations are the following: leg segments and feet (L1), leg joints (L2), leg segments (L3), leg joints and feet (L4), feet (L5), pelvis, leg segments and leg joints (L6), leg segments and joints (L7), pelvis and leg joints (L8), and pelvis (L9). Fig. 10 (right) shows the number of works focused on each of these groups (the total number of works is 90).

In the lower half body, it is noticeable that most of the works focus on the leg joints, L2, which is the object of monitoring most works (63/90 studies). These joints are commonly studied in multiple applications, being the most important the gait analysis because of their relevance in health assessment. It is worth mentioning the contrast of the number of studies about the L2 group in comparison with the L4 group, which combines the leg joints with the feet, meaning that the motion of foot joints is commonly discarded in the studies focused on the lower limb joints. Besides the leg joints, the following most studied groups that are also focused on legs are leg segments, L3 (8/90 studies), leg segments and joints, L7 (7/90 studies), and leg segments and feet, L1 (5/90 studies). The rest of the groups, which include leg joints and feet (L4), feet (L5), pelvis, leg segments and leg joints (L6), and pelvis (L9), are studied in few works (2/90 studies).

The monitored units in these groups are measured regarding their orientation or location. Orientation refers to the rotation angles, which are commonly presented as Euler angles or quaternions. Locations refer to the spatial coordinates, so they are a measurement of distance. Most studies estimate the orientation of monitored units (81.6%), 15.6% give a combination of orientation and location of units, and only 2.7% are focused only on providing locations. Fig. 9 shows the distribution of the measured magnitudes.



Fig. 10. Anatomical monitored units. Left: location of the monitored units in the upper or lower half part of the body. Right: number of papers per combination of segments and/or joints monitored.

#### D. Adopted Algorithms

The algorithms used in the estimation of the kinematic parameters can be separated into five different groups: integration, vector observation, sensor fusion filters, ML techniques, and other methods.

1) Sensor Fusion Filters: Sensor fusion filters (SF in Table III in Appendix A), including KFs, particle filters (PFs), and CFs, are the algorithms most frequently used. Specifically, KFs are still the algorithms that are employed the most in the inertial human motion analysis field, following the trend reported in previous studies [4].

The problem formulation of Bayesian filters consists of the identification of the desirable estimations using a series of measurements observed over time containing statistical noise and different inaccuracies [156]. The inputs and observations form the knowledge of the system's behavior both convey errors and uncertainties, namely, the measurement noise and the system errors. These filters fuse the information of sensors with the knowledge of the system in two stages: the estimation stage and the update stage. The initial stage uses the information of the previous time instant to estimate the current state of the state vector. The second stage updates these estimations using the measurements from the sensors.

The motion analysis includes proposals with extended KF (EKF), KF, and unscented KF (UKF) in descending order of frequency of use. KF is a sensor fusion technique that estimates the states of a linear system through the minimization of the variance of the estimation error [156]. KFs use a series of measurements observed over time and their statistical noise to produce estimates of unknown variables. EKFs appeared because KFs are limited to linear systems, being their generalization to nonlinear systems. EKFs assume that the nonlinearities in the dynamic and the observation model are smooth, so they expand the state and observation functions in the Taylor series and approximate, in this way, the next estimate of the state vector. However, this approximation can introduce large errors in the true posterior mean and covariance of the variables, which may lead to the divergence of the filter. One of the possible solutions is the use of UKFs, whose distribution of their state vector is a set of sample points called sigma points. Sigma points capture the actual mean and covariance of the Gaussian random variables and are obtained though the *unscented transformation* (UT). The UT is a method for calculating the statistics of a random variable that suffers a nonlinear transformation. UKFs are an extension of UTs to the recursive estimation where the UT is applied to the augmented state vector.

EKFs are the KF variation that most frequently appears in motion analysis. EKFs are used for the sensor fusion of gyroscopes and accelerometers in order to estimate the joint orientation [8], [38], [66], [70], [72], [92], [96], joint orientation and location [55], [76], [79], [88], [108], and segment orientation [77], [78]. Researchers also use EKFs to fuse gyroscope and accelerometer data with magnetometer measurements to estimate the orientation of joints [138], [142] or segments [117], [118], [135], [140], [145]. EKFs are also combined with the Gauss–Newton algorithm to fuse the information of gyroscopes and accelerometers, and estimate the orientation of joints [64].

Classical KFs are normally used for the sensor fusion of gyroscope and accelerometer data to estimate the segment orientation [33], [40], [43], [50], [52], [57], [97], [139], the joint orientation [44], [82], [85], [95], [98], the segment location [74], and all of them, the segment and joint orientation and location [42]. KFs have also been used in the fusion of gyroscope, accelerometer, and magnetometer data to estimate segment orientation and location [136], [152].

UKFs appear less frequently in the literature. UKFs are commonly used for the sensor fusion of gyroscopes, accelerometers, and magnetometers to estimate the joint location [137], or orientation [130], [149], the orientation of joints and segments [131], and the orientation and location of both elements, joints and segments [121]. Some works do not use the magnetometer information and only fuse the gyroscope and accelerometer data to estimate joint orientation [99], [106].

PFs are another modification of KFs for their use in nonlinear systems [156]. PFs are close in functioning to UKFs but with a set of differences that approximate PFs to a generalization of UKFs. PFs update the estimations with a randomly generated noise according to the *prior* knowledge of the process noise probability density function (pdf) instead of the update of the UKF that is deterministic. Another difference with UKFs is that the number of particles in PFs is not related

to the length of the state vector. Finally, PFs estimate the pdf of the state instead of the mean and covariance, and it converges to the actual pdf as the number of particles increases.

PFs are less popular than any other kind of KFs. PFs are applied to fuse the measurements from gyroscopes, accelerometers, and magnetometers to estimate the orientation of segments and joints [125]. PFs are combined with other KFs, such as EKFs, for the fusion of gyroscope and accelerometer measurements to estimate the orientation of segments [94].

CFs combine the information from different sensors by minimizing the mean square error instead of the error covariance, which is minimized in KFs [157]. CFs are used to fuse the measurements of gyroscopes, accelerometers, and magnetometers to estimate the orientation of joints [123], [126], [133], their orientation and location [119], [144], and their orientation together with the segments orientation [143]. They are also used to estimate the joint either by combining the information of the gyroscope and the accelerometer orientation [45], [47], [62], [65], [101], [109], or just only with the gyroscope [11].

Another alternative to KFs is the weighted Fourier linear combiner (WFLC) filter, which is a model-based adaptive filter. WFLCs exploit the *prior* knowledge of the signal shape and evolution over time, in those occasions when the motion performed is given [93]. These filters are especially effective in periodic signals but adapt to variations between repetitions. Their applications to the human motion analysis include the use of the turn rate measurements to estimate the segment orientation [14] and the combination of these data from the gyroscope with the accelerometer measurements to estimate the orientation of joints [93].

2) Data Science Algorithms: ML techniques represent the second group of algorithms that are applied most frequently for the estimation of human kinematics. Furthermore, supervised learning algorithms are the most widespread in recent years. Supervised learning is one of the most employed learning paradigms, which tries to discover the unknown function  $f(\mathbf{x}, \omega)$  that relates the input space  $X \subset \mathbb{R}^n$  (which, in this work, are the inertial measurements), with the output space  $Y \subset \mathbb{R}$  (which describes the motion kinematics). Each pair  $(\mathbf{x}_i, y_i)$  is composed of the value of a set of n predictive variables  $\mathbf{x}_i = (x_1, \ldots, x_n)_i \in \mathbb{R}^n$  of the input space, which is measured by the IMUs, and its corresponding output value  $y_i \in \mathbb{R}$ , which are the target value of joint or segment orientation and location. During the process called training, supervised algorithms retrieve the map  $f \in F$ from the provided training dataset  $\mathcal{D}$ , typically establishing an optimization problem that minimizes a loss function  $\mathcal{L}$ . Different parametric function spaces F with different learning methods correspond to the existent variety of supervised methods, as described in depth in Table II.

Gaussian processes (GPs) are kernel-based probabilistic ML models. The GP is a kind of continuous random process f(t) such that every finite set of random variables has a multivariate Gaussian distribution [158]. The GP method estimates the output y by introducing a set of latent variables  $\{f(t_k)\}_{k=1}^n$  from a GP and explicit link functions,  $g(\cdot)$ . GP latent variable

(GPLV) models are used with the gyroscope and accelerometer data to estimate the segment positions [58].

Other classical ML methods are decision trees (DTs) and support vector machines (SVMs). A DT is a classical ML method that builds a tree, a particular graph without cycles, by branching decision paths for each considered input variable to make the final classification [159]. During the training process, databases are used to compute thresholds (the parameters in DTs) that better branch the input variable for optimizing a criterion, usually the best gain of information possible in the current node (optimizing the entropy), for a better prediction of the output variable.

SVM is one of the most used ML methods for classification [160], [161]. It establishes an optimization problem to find the so-called *support vectors*, those training data that are close to the separation hyperplane, and maximize the *soft margin*. Frequently, this method uses the kernel trick that consists of choosing an appropriate nonlinear mapping  $\phi$  that maps input samples into a higher dimensional space where they are likely to be linearly separable. In regression problems, the support vectors are used to provide a continuous value through a link function instead of classes.

However, these classical ML methods are less promising than artificial neural networks (ANNs) in the human motion analysis field, as proved in [109] for the correction of the joint angles initially obtained from sensor fusion filters. ANNs consist of a set of connected base units known as artificial neurons that emulate the biological neurons of animal brains [162]. ANNs are usually organized in layers, which interconnect themselves to create a huge variety of networks that try to represent the functional relation between the input and output variables. ANNs have revolutionized the ML field due to their ability to model very complex nonlinear input–output relations and their capacity to learn them from a huge amount of data. The single-multilayer perceptrons (MLPs) were the first ANNs.

In the inertial motion capture field, ANNs use the accelerometer data as inputs to estimate the segments' orientation and location [27], combine the gyroscope and accelerometer data to estimate joint orientations [68], [71], [84], [91], or fuse the information of the three sensors integrated into IMUs to estimate the segment angles [147]. Other specific types of ANNs merge the estimation of joint angles with gyroscopes and accelerometers, such as the general regression NNs [49], [90] or the Elman neural networks [115].

Deep neural networks (DNNs) arise later than ANNs and encompass a huge amount of modern network architectures with a high number of interconnected layers [163]. The current technology allows massive computation during the training process and, hence, a new variety of interconnections and predictions in real time. DNNs start with the convolutional neural networks (CNNs), a large sequence of convolutional layers configured in a cascade where each layer computes the convolution operation (see [164]) from the previous one. They are able to extract intrinsic local features, called *deep features*, which surpass the results of the classical ML methods. VGG [165] and residual networks (RESNETs) [166] are famous CNNs included in this category. Most of the DNNs including CNNs are feedforward networks, which means that the information flows forward, and they do not include cycles. However, DNNs also include recurrent networks, which memorize internal states, frequently exploited for temporal sequences, such as the improved recurrent neural network (RNN) [167], which evolved to the novel long short-term memory (LSTM) [168], the gate recurrent unit (GRU) [169], and the nonlinear autoregressive neural network with exogenous inputs (NARX) [170].

Among the deep learning algorithms, LSTMs are the most utilized. LSTMs are made by a sequence of cells capable to keep previous states and specifically keep two kinds of temporal information and the LSTM. They have replaced the RNNs that suffer from the vanishing gradient problem during the training and include forget gates to quickly adapt to the new changes of data. These DNNs can use just the information of accelerometers and the orientation of a set of body segments to estimate the whole-body posture [29] or fuse the information of specific force with the turn rate to estimate the joint angles [53], [69], [86], [103]. A less common approach includes the fusion of gyroscopes and magnetometers to estimate the joint angles [155]. LSTMs can also be used to estimate the orientation of the whole-body joints using the orientation obtained with sparse commercial sensors [148]. In [69], LSTMs are combined with CNN to estimate the joint angles. CNNs are also used to obtain the joint angles only using the accelerometer data [21] or fusing the gyroscope and accelerometer data [56], [111], [114]. Mundt et al. [34] made a comparison of these previous methods, CNNs and LSTMs, together with MLPs for the estimation of joint orientation. Using the information from gyroscopes and accelerometers, CNNs provided the most favorable metrics. Other RNNs are also used to estimate joint orientation. To estimate the joint angles from gyroscopes and accelerometers, Tham et al. [54] propose an NARX; Conte Alcaraz et al. [134] also include the magnetometer data with NARX and LSTMs.

The ML-based algorithms that are used for human motion analysis are supervised methods. These methods need training data, which must include reference data of the parameter to estimate, i.e., the joint or segment orientation or location. In this review, we found 26 works that use reference data, which can be obtained from a stereophotogrammetric system (17/26) [21], [27], [34], [49], [53], [54], [56], [58], [68], [69], [86], [90], [91], [103], [111], [114], [115], electrogoniometer and encoders (2/26) [71], [84], or inertial sensors (7/26) [29], [100], [109], [134], [147], [148], [155]. Fig. 11 shows the percentage of use of the different external sensors for obtaining reference data.

The use of optical systems also allows data generation in order to have data for the training and testing of the algorithms and to increase the available data to train the models. These tasks can be performed with simulation software, e.g., Open-Sim [171], as in [86], [103], and [148], by applying kinematic relationships from the stereophotogrammetric measurements, as in [19], [53], [56], [68], [69], [91], and [103], or with data augmentation techniques [68].



Fig. 11. External sensors used to obtain the reference measurements for the training and testing stages of the ML algorithms.

*3) Other Algorithms:* Over the years of research on motion analysis with inertial sensors, proposals have been based on various algorithms other than sensory fusion filters and data science methods. These proposals cover the integration of the gyroscope data to estimate the joint orientation [12], [15], [16], [17], [41] to its combination with the direct use of the data from accelerometers to estimate the orientation of joints [80], [104], [105] and segments [59], [60], [63], [81], and to estimate the orientation and location of segments [110], [112]. The measurements from gyroscopes and accelerometers are also used directly to obtain the orientation and location of joints [35] and segments [46], and to estimate the orientation of both joints and segments [48], [51], [113]. The information of the three sensors in the IMU is also directly used for the estimation of the segments orientation [141].

Different works exploit the observation of the gravity vector by the accelerometer for the estimation of the orientation of joints [18], [22], [24], [30], [61], segments [28], or both [20]. Other works also use the data of the magnetometer to estimate the joint orientation and location [31] or combine this information with the measurements of accelerometers to obtain the joint orientation [153]. The gravity vector can also be observed by eliminating the linear acceleration of the motions, which can be estimated from the turn rate measurements [73].

Besides the gyroscope integration and the direct observation of vectors, the measurements from IMU sensors can be combined through the use of virtual sensors. The use of virtual sensors consists of the estimation of the measurements that a sensor would obtain if it was located in the joint. This measurement projection is commonly performed because it is not possible to place the sensors in the joints. This is commonly used to simulate the measurements in joints, whereas the IMUs are placed in segments. This approach is used to combine the gyroscope and accelerometer measurements to estimate only the joint orientation [37], [46], [83] or both the joint orientation and location [89]. Another application of virtual sensors is to combine the turn rate, specific force, and magnetic field to obtain these magnitudes in joints to estimate their orientation [129], [132], [150], [151] and also not considering the measurements from gyroscope for the joint orientation estimation [154].

Another common method used is the gradient descent. Gradient descent is applied to obtain the joint orientation by using the measurements from gyroscopes [13] and combined

## TABLE II

OUTCOMES OF THE ANALYSIS OF INPUTS USED IN THE ML ALGORITHMS, ALGORITHMS APPLIED IN EACH WORK, AND OUTPUTS AIMED AS TARGETS WITH THE CAPTURE SYSTEM EMPLOYED. SF: SPECIFIC FORCE; TR: TURN RATE; OR: ORIENTATION; CAP.: CAPTURE; SNN: SHALLOW NEURAL NETWORK; DNN: DEEP NEURAL NETWORK; AND VS.: VERSUS

Work		Input	s	Specifications of inputs	Algorithm	Output	5
WOLK	SF	TR	OR	Specifications of inputs	Algorithm	Magnitudes	Cap. system
[90]	<u>↓</u> <u>√</u> _	L <u>-</u> _	L	Time instant	GRNN	_Joint angle	Optical
[49]	$\checkmark$	✓		Time instant	GRNN Aux Sim Info	Joint angle	Optical
- [84]	+				$ -\overline{ANN}  =$	Joint angle	$  \overline{Other}$ $  -$
[58]	†	$\left  \frac{1}{\sqrt{2}} \right $		Sparse IMUs	GPLVM	Whole-body posture	Optical
	+			$\vec{s}$ parse $\vec{I}$ $\vec{M}$ $\vec{U}$ $\vec{s}$		Segment location	
[147]	↓ <b>~</b>	L	<b>↓</b>	5 samples (0.55 s) windows	SNN & DNN	Whole-body posture	Inertial $      -$
[109]		✓	~	Joint angle from Bayesian filters	NN (vs.) SVM (vs.) DT	Joint angle	Inertial
[53]	✓	✓		Simulated inertial data Different IMU combination + PCA From 100 to 1000 time-steps	LSTM	Joint angle	Optical
[21]	<b>↓</b>			600 ms windows Sparse IMUs for leg kinematics	1D-CNN	Joint angle	Optical
[56]	+   ✓	 ✓		Experimental and simulated data combined 100 time-steps for each walking cycle	CNN	Joint angle – – – – Joint moment GRF	Optical
[27]	 ↓ ✓			Time instant	ANN	Joint angle Joint moment GRF	Optical
[68]	   ~			Simulation of IMUs Data augmentation 10 frames windows Full gait cycles	ANN	Joint angle Joint moment	Optical
[91]		 _ ✓ _		Simulated inertial data Segmentation into gait cycles	ANN	Joint angle	Optical
[29]	$\checkmark$		√1 <b>x</b>	Sparse IMUs Window of 50 samples $= 0.50$ s	LSTM	Whole-body posture	Inertial
[69]	   √			Simulated inertial data Calibration for alignment Marker trajectories filtered 1 s & 2 s windows	1D-CNN (vs.) LSTM & Optimizer	Joint angle time series	Optical
[86]	† ✓	 ✓		100 samples (100 Hz) windows	- <u>C</u> NN	Joint angle	Optical
[103]		 ✓		Experimentally measured IMU data, Simulated inertial data 200 time-steps Time wrapping of strides as inputs	LSTM	Joint angle time series	Optical
[155]	↓	−		Sparse IMUs	LSTM	Joint angle time series Whole-body posture	Inertial
[148]			~	+Joint parmeter Sparse IMUs 300 time-steps windows	LSTM	Whole-body posture	Inertial
[53]		✓		Experimental and simulated data 101 frames windows	- <u>ML</u> P LSTM CNN	Joint angle time series	Optical
[71]				Time instant	ANN	Joint angle Walking speed	Other
[54]	✓	<ul> <li>Image: Construction of the second seco</li></ul>		Time instant	NARX (vs.)   CF	Segment angle	Optical
[134]				Sparse IMUs for leg kinematics Gait cycle segmentation	NARX	Joint angle	Inertial
[100]	✓	✓		Sparse IMUs for knee angle Short intervals 0.05 s	LSTM	Joint angle	Inertial
[111]	†	 ✓		Sparse IMUs for leg kinematics	 	Joint angle	Optical
[114]	IV	[]		Initial gyro integration for joint orientation estimation	Elman NN	Joint position	Optical
[115]	✓	√		Gait variations	LSTM	Joint angle time series	Optical

with the specific force measurements [39], [64], [102]. This approach also allows the gathering of the measurements of the three sensors to estimate the segment orientation [116].

The remaining proposals use a wide variety of methods and approaches to monitor the measurement units. These methods include probabilistic graphical models [122], which



Geometrical

Fig. 12. Scheme of the BCs commonly implemented in the kinematic models for the inertial motion analysis. It includes the geometrical model with a reduced number of DOF in the knee and ankle joints, and the limitations with respect to the ROM using the knee as an example. The lengths and the IMU-joint vector used in the soft constraints are labeled as  $d_S$ ,  $d_T$ , and  $\mathbf{r}_i$  with i = 0, 1, 2, respectively.

are employed with the gyroscope, accelerometer, and magnetometer measurements to estimate the orientation of joints, smoothing algorithms [36], bidirectional low-pass filters [19], least squares [75], [87], optimization techniques [25], [67], [107], and modified iterative algorithms [23]. The latest worth mentioning approach consists of the double-sensor differencebased algorithm [26], which combines the measurements from two accelerometers placed on the same segment with the knowledge of their positions with respect to the joint.

4) Approaches for Error Reduction: This section describes the main approaches found in the literature in order to reduce the errors in the estimation of kinematic parameters. First, we focus on the explanation of the proposals based on BCs, and then, we summarize the approaches for error reduction based on the properties of the inertial sensors and their motions.

BC is a promising resource to improve inertial human motion analysis. A common approach is to model the rotations of different joints with different degrees of freedom (DOF), which are depicted with cylinders in the geometrical model in Fig. 12. The three rotational DOFs recorded by IMUs can be modeled as one or two DOF joints, according to the possible anatomical motions.

In the literature, the knee is assumed to be a hinge joint with just one DOF (flex-extension; see Fig. 12) [8], [39], [45], [55], [62], [71], [80], [87], [88], [96], [101], [130], [135], [136], [151] because its internal rotation allows a negligible range of motion (ROM). Some works also consider the knee's internal rotation together with its flex-extension, so the modeled joint presents two DOFs [82]. The same approach can be used to simplify the ankle orientation estimation, using only the knee-flex extension rotation, so one DOF [136], or also including the internal rotation, resulting in two DOFs [55], as depicted in Fig. 12. Despite being less commonly used due to its actual DOF and complexity, hips are modeled as joints with two DOFs [101], [151].

This DOF reduction can be also applied to upper limbs. Elbows can be assumed to have only two DOFs, gathering the elbow flex-extension rotation and the forearm internal-external rotation [13], [55], [98], [99], [106], [121], [122], [143], [149], [150] or considering only the elbow flex-extension [125]. The wrist can be also modeled as one DOF joint [121], [122] or allowing a second rotation with two DOFs [150].

Another approach related to the simplification of motions to a lower amount of DOF is to model the motions as if they occur in one or two planes. This approach reduces the 3-D space in, at least, one dimension. Different motions, such as gait or squats, can be approximated as 2-D in the sagittal plane [105], [109], [134] or with the combination of the sagittal and coronal planes [52], [97].

The separation of motions in the DOF available for the joints allows another restriction based on the joint anatomical ROM. This constraint is based on the correction of the estimations that are not consistent with the anatomically possible ROM per DOF of joints. As depicted in Fig. 12, the ROM of a joint, in this case, the knee, includes the consistent estimations and the estimations on the limit of the ROM. The values of angles outside this range are wrong estimations of the algorithm, and the objective is to detect and correct them. This approach can be found in several proposals in the literature [59], [79], [87], [120], [136], [143].

Another common approach for the error reduction in most of the biomechanical models is to take into account the anatomic parameters, such as the joints location with respect to the sensors or the segments length [8], [17], [19], [20], [23], [24], [25], [26], [35], [37], [38], [46], [46], [64], [73], [76], [77], [78], [79], [83], [87], [88], [89], [91], [92], [93], [94], [95], [96], [98], [99], [105], [106], [107], [108], [119], [129], [132], [137], [138], [143], [150], [151], [154]. Fig. 12 shows these parameters, labeled as  $r_i$ , where  $i = 0, 1, 2, d_s$ , and  $d_T$ , respectively. The IMU-joint vector and the segment length are used to impose that the relationship between magnitudes has to be consistent with the anatomy of participants and the location of sensors on the body. These constraints can be applied by the relationship between the velocity in the common joint between two segments and the turn rate measured by the gyroscopes of the corresponding IMUs using (1) (see Fig. 12)

$$\boldsymbol{v}_{knee} = \boldsymbol{\omega}_{IMU_1} \times \boldsymbol{r}_1 = \boldsymbol{\omega}_{IMU_2} \times \boldsymbol{r}_2. \tag{1}$$

An alternative approach is to relate the linear acceleration suffered by the IMUs with the linear acceleration in the common joint between segments. This approach requires the consideration of the gravity influence from the specific force measured by accelerometers. If the gravity influence is eliminated, (2) can be applied with the derivation of the turn rate

$$\boldsymbol{a}_{knee} = \boldsymbol{a}_{IMU_1} + \dot{\boldsymbol{\omega}}_{IMU_1} \times \boldsymbol{r}_1 + \boldsymbol{\omega}_{IMU_1} \times (\boldsymbol{\omega}_{IMU_1} \times \boldsymbol{r}_1)$$
$$= \boldsymbol{a}_{IMU_2} + \dot{\boldsymbol{\omega}}_{IMU_2} \times \boldsymbol{r}_2 + \boldsymbol{\omega}_{IMU_2} \times (\boldsymbol{\omega}_{IMU_2} \times \boldsymbol{r}_2). \quad (2)$$

The IMU-joint vectors combined with the segment lengths or the joint–joint vectors are commonly used to estimate the kinematic of chains of segments. This is frequently performed with the Denavit–Hartenberg (D-H) notation, which uses four angular and distance parameters to relate reference frames with the links of spatial kinematic chains [172]. In order to apply the D-H convention, one reference frame is defined for each DOF included in the biomechanical model. The axes of the consecutive reference frames, i - 1 and i, must follow two rules: the  $x_i$ -axis must be perpendicular to  $z_{i-1}$ , and the  $x_i$ axis must intersect with  $z_{i-1}$ . In this way, the transformation matrix  $T_{i-1,i}$  detailed in (3) defines the transformation between consecutive frames

$$T_{i-1,i} = \begin{bmatrix} \cos \theta_i & -\cos \beta_i & \sin \theta_i & \sin \beta_i & \sin \theta_i & r_i \cos \theta_i \\ \sin \theta_i & \cos \beta_i & \cos \theta_i & -\sin \beta_i & \cos \theta_i & r_i \sin \theta_i \\ 0 & \sin \beta_i & \cos \beta_i & d_i \\ 0 & 0 & 0 & 1 \end{bmatrix}$$
(3)

where  $\theta_i$  is the angle between the  $x_{i-1}$ - and  $x_i$ -axes, about the  $z_{i-1}$ -axis, and  $\beta_i$  is the angle between the  $z_{i-1}$ - and  $z_i$ -axes, about the  $x_i$ -axis. This transformation of consecutive frames allows the estimation of the forward kinematics of a chain of joints by using (1) and (2), as performed in [74], [79], [87], [121], and [122].

The location of joints and the segment lengths are not imposed as the limitation of DOF or ROM, which directly models or corrects the estimations. However, they are used to impose constraints on the measured magnitudes. For that reason, the restrictions forced through these conditions are commonly known as *soft constraints*. It is worth mentioning that the errors in the estimation of the IMU-joint vector directly influence the estimation of joint angles that use these soft constraints [87].

OCs found in the literature that are not related with biomechanical properties include the following: modeling the bias of sensors or including them in the state vector [13], [19], [30], [33], [36], [44], [46], [52], [57], [60], [61], [62], [63], [70], [75], [82], [83], [95], [117], [118], [119], [123], [125], [139], [153], [173]; calibrating this bias [30], [44], [46], [97], [104], [105], [121], [124], [133], [150]; low-pass filtering the recorded signals [11], [14], [37], [41], [45], [63], [71], [143], [145]; or optimizing them [67], updating the estimations when a direct observation of gravity is available or with its dynamic compensation [35], [43], [48], [51], [59], [81], [140], [141], [174] or the zero-angle during the zero-turn rate time instants [11], [47], [48]; modeling the disturbances in the magnetometer and gyroscope [31], [118]; using virtual sensors, discriminating the quasi-static and dynamic motions [41]; eliminating the errors from the soft tissue artifacts [44], [127]; and, in the case of KFs, the optimization of the Kalman parameters [33], [50], [79].

## E. Participants of the Study

This work also analyzes the NS that participate in the studies to validate the methods. Fig. 13 (top) shows the *boxplot* of the distribution of subjects in the studies analyzed in this work (147). The boxplot presents the first, second, third, and fourth quartiles of the studied subjects together with the *outliers*. In this work, the outliers represent the punctual studies that test their proposals in more than 18 subjects (8/147 studies). According to Fig. 13 (top), most results provided in the studies correspond to a population of fewer than ten subjects.



Fig. 13. Characteristics of the population of study in the human motion analysis literature. Top: percentage of works that evaluate their proposal in each number of participants. Bottom: percentage of works that consider a population with (diseased vols.) or without (healthy vols.) disease.

It is worth mentioning that the median is 3 subjects per study, which makes the results hardly generalizable to all populations. Furthermore, more than one-third of studies test their proposals with only one person (34.7%).

The studies that validate their proposal with the highest amount of volunteers commonly test ML-based algorithms. This amount of volunteers is required by these algorithms because they work with a high amount of data in order to develop generalizable models. However, 65.4% of these studies use the data from the optical systems to generate simulated inertial data, as shown in Fig. 11.

Studies analyze volunteers with or without diseases related to the motor system. We assume that, if there is no statement about whether unhealthy people are included, the studied population is healthy or with no illness that affects the performance of motions. In this way, only a small percentage (8.2%) of proposals are tested on population with these motor limitations [see Fig. 13 (bottom)]. This is remarkable since most proposals claim healthcare applications among their possible uses, as seen in Section III-B (95.2%).

# F. Validation Systems and Evaluation Metrics

For the validation of the reviewed works, researchers use different systems, as shown in Fig. 14. The gold standard is the 3-D optical motion capture system, such as the



Fig. 14. VS used to assess the proposals. Force platforms and simulations are abbreviated as force plat. and simul., respectively.

commercial Vicon [175] or Optitrack [176], being the most widely employed. This system is commonly used for the validation of proposals (68.0%) and, sometimes, (4.8%), in the combination of force platforms or with simulation software (0.7%). 2-D optical systems that can obtain a reference in the image plane are also used (4.8% of the works). In some works, the 2-D optical systems are combined with depth sensors (0.7%).

Another approach to validate the algorithms that is worth mentioning is the use of output values of commercial inertial systems that provide highly accurate measurements, as done in the 6.8% of studies. IMUs of the Xsens commercial brand are the most frequently used for the validation of proposals [177]. Four of the eleven works that validate their algorithms against inertial sensors outputs use these sensors [109], [110], [147], [148], whereas the remaining seven works use IMUs of seven different brands.

A less common solution includes the use of analog and electronic goniometers (8.2%). Other VSs, which, in combination, sum 6.8% of works, include different programs of motion simulation, encoders, and potentiometers.

The accuracy metric reported most frequently for the validation of proposals is the RMSE. In some works, the CFC or the MAE is also provided. For the case of angles measurement, the studied works report an RMSE between  $2.59^{\circ}$  and  $7.67^{\circ}$ . Even if, on average, most of the studies provide similar metrics, it is worth mentioning that the RMSE range of ML methods is between  $2.48^{\circ}$  and  $5.70^{\circ}$ , whereas the RMSE provided by the classical methods is between  $2.24^{\circ}$  and  $7.80^{\circ}$ . These results prove that ML methods are promising approaches in the human motion analysis field in spite of their limitations related to data availability.

# IV. DISCUSSION

After the previous in-depth study, we discuss the review findings in terms of the general trends and the future guidelines in the inertial motion monitoring field.

# A. General Trends

This article analyzes the current state and the research trends in the inertial motion analysis field. The analyzed works show interest in developing an alternative to the *gold standard* system, based on cameras, due to their high cost

and the required space of use. As a consequence, the research focused on developing an IMU-based system for human motion analysis has increased over the last years, as seen in Fig. 5.

The sensors integrated into IMUs, accelerometer, gyroscope, and magnetometer are fused in different ways in the analyzed works, as shown in Fig. 6. The fusion of gyroscopes and accelerometers is more common than the use of magnetometers, being one of the main differences with respect to the findings in previous reviews [4]. However, the use of magnetometers is spreading during the last year with ten works. Bayesian filters and trigonometric approaches are the ones that most frequently employ the data from the magnetometer. ML proposals barely rely on the measurements of the magnetic field and only use them in the training step of the algorithms. In this way, the works focused on ML proposals are not limited by magnetic disturbances.

The common objective in 72.4% of studies (see Fig. 7) is to obtain 3-D kinematic parameters. The 2-D estimations are useful in human motion analysis because some movements can be simplified as motions in one plane, e.g., knee or elbow flex extension or even gait and squats. However, these estimations can miss relevant information in motions, about correctness or symptoms of motion-related diseases. Obtaining complete kinematic information is especially important in healthcare applications, which is considered in the 95.2% of works (see Fig. 8).

Most of the analyzed works (33.5%, as shown in Fig. 8) mention the generic motion capture field for human motion analysis, closely followed by the gait evaluation, as aimed applications for their work. That reflects the interest in developing more affordable and user-friendly alternatives to optical systems, as previously discussed. Consequently, the analyzed works propose algorithms to monitor frequently the orientation of joints, which is commonly measured by stereophotogrammetric systems, such as Vicon [175]. 64.6% of works focus on joints and 81.6% on the estimation of the orientation. These percentages imply a great advance in the direction of inertial solutions for human motion analysis, especially compared to the trends reported in [4], where most works studied the orientation of segments.

Another interesting analysis is focused on the distribution of works with respect to the analyzed body part, divided into the upper and lower halves. 61.2% of works (see Fig. 10) study the lower half of the body, and most of them focus on the leg joints, which is also consistent with the trend of gait evaluation besides the motion analysis. Conversely, only 34.7% (see Fig. 10) study the upper half, which includes arms and trunk, which are difficult to monitor due to the DOF and the complexity of joints as shoulders or neck. The results in Fig. 10 mean that, during the last years, the research has been focused mostly on the lower half of the body, the opposite of what happened in [4], and most of the reviewed works analyzed upper limbs. However, monitoring this upper half of the body is crucial for the evaluation of motions, being especially important in the rehabilitation of cognitive alterations or illnesses, such as strokes. The remaining 4.1% (see Fig. 10) of proposals are aimed at monitoring the wholebody posture, which is the most complete approach for the human monitoring. Even though the gait analysis is commonly performed by monitoring the lower limbs, the upper limbs are also important to study relevant features, such as balance, in clinical assessments. The rising interest in considering the full body is also another noteworthy difference compared to previous findings.

For full-body monitoring, ML methods are especially attractive. Sensor FAs use one IMU per segment to monitor the whole-body posture or model biomechanical relationships between segments to reduce this number with different constraints. Conversely, the approach in ML-based proposals focused on the whole-body posture is the optimization of the number of devices with the use of the so-called *sparse* IMUs, as in [29], [58], [147], [148], and [155]. This approach is also used to monitor specific limbs, such as legs, reducing the number of sensors [21], [100], [134].

The biomechanical approaches for error reduction can restrict motions and might not be generalized for populations with motor-related diseases. For instance, the ROM of joints can be different in people with anomalous physical abilities. Likewise, the assumption of a number of DOF can miss relevant information about motions out of the main directions. Also, the knowledge of the segment length or the location of the sensors on the body is not always available in practical applications. Different IMU-joint calibration methods have been proposed to address this limitation. The first approach is to obtain an average location of joints with respect to the sensors, which has been validated for the upper and lower limbs [178], [179], respectively, but requires specific calibration motions. The second method consists of estimating an adaptive position vector, considering the changes in the location of IMUs due to soft tissue artifacts [180], [181], [182], [183], which has been validated for the calibration of hips performing leg circles [182]. These proposals assume that the joints are fixed, but it is not the case in all the activities in daily living. In [184], the calibration of moving joints with soft tissue artifacts is addressed.

As in previous findings [4], the sensor FAs are employed more commonly than other approaches. However, their use during the last decade remains stable (around 24 papers), whereas the use of ML techniques has increased from four papers to 18 papers. These ML techniques provide a slight improvement in the accuracy metrics, referred to as a reduction of the maximum RMSE in  $2^{\circ}$  in angle measurements. However, none study analyzed in this work includes research that makes a fair comparison using common data to test both approaches.

Sensor fusion filters and data science algorithms differ in terms of their computational costs. Computational time varies among the different methods depending on their implementations. Sensor fusion filters are faster than data science methods, which require more calculations, especially DNN-based ones whose number of parameters is superior. Also, ML- and DNN-based methods usually demand more memory than the sensor fusion solutions, especially in their training stage, making their implementations more expensive.

ML algorithms are more robust to variation in the intrinsic noise of the sensors with which they are trained. In addition, their robustness can increase by generating synthetic data to which more noise models are added. Conversely, the sensor FAs include parameter tuning to adapt them to the sensors used, e.g., the covariance matrix of KFs. Thus, sensor FAs would require a previous study to estimate these sensordependent matrices.

ML methods require a high amount of reference data to be trained. Two alternative trends are followed in order to generate reference data: 1) to simulate the inertial data from the optical data to use them as inputs or 2) to use the orientation data obtained by commercial systems as reference. In the first case, the simulation of inertial data might not present the intrinsic errors of IMUs, whereas, by using inertial data as reference, it presents an error around  $0.5^{\circ}$  depending on the commercial brand, which is less accurate than optical systems.

With regard to the validation, new reference systems have appeared during the last few years. Among them, we find 2-D visual systems, encoders, and computational models. The 3-D optical systems are still the ones most frequently used (68.0% of studies; see Fig. 14). However, the use of this VS entails the limitation of testing the proposals in daily activities and alternative validation methods should be investigated [7].

The reviewed studies generally analyze a low amount of participants for the validation of the algorithms. This limitation was detected in [4] and still remains in recent works. Most studies test their results only on one volunteer, and the average of study subjects is four participants. It makes the proposals hard to generalize for the whole population. In those studies in that more participants are involved, the inertial data are simulated from the optical data, or their reference consists of the orientation outputs obtained from the IMUs, including the errors previously indicated.

Most studies analyze healthy participants. That is noticeable since most studies consider healthcare applications as possible uses of their proposals. However, only a few of them (8.2%) test their proposals on subjects with motor-related diseases.

## B. Future Advancements and Developments

This review highlights a set of clear trends. The studies describe the motions in the 3-D space more frequently than reducing them to planar motions. This is crucial to describe complex motions that can be performed during daily life, so it is required for an out-of-the-lab analysis. Furthermore, the reduction of the gait or simpler motions, such as knee flexextension to a plane, eliminates relevant information about these motions.

The reviewed works focus on the lower limbs, specifically on the orientation of the hip, knee, and ankle. Future research should include upper limbs or even focus on the development of whole-body posture monitoring for a complete description TABLE III

FUSION; ML: MACHINE LEARNING; AND OA: OTHER ALGORITHMS. OTHER ACRONYMS; BC: BIOMECHANICAL CONSTRAINT; ANT: ANATOMICAL PARAMETERS IN USE; OC: OTHER CONSTRAINTS; ML LEARNS: REFERS TO THAT THESES ML-BASED ALGORITHMS LEARN FOR REAL MOTIONS, SO THEY ARE CONSTRAINT TO ANATOMICAL JOINT LIMITS; EST: REFERS TO THE USE OF ANATOMICAL OR SOFT CONSTRAINTS AFTER ESTIMATE THE REQUIRED PARAMETERS; GS: GYROSCOPE SENSOR; AS: ACCELEROMETER SENSOR; MS: MAGNETOMETER SENSOR; OS: OTHER SENSORS FOR TRAINING; RELEVANT DETAILS RELATED TO THE IMPLEMENTED ALGORITHMS, THE SENSORS IN USE, AND THE ESTIMATIONS OF THE SELECTED STUDIES. WITH RESPECT TO THE FA ACRONYMS, SF: SENSOR MENT

(7)
m
$\overline{\mathbf{x}}$
÷
<u>-</u>
G
S
$\sim$
Ħ
2
~
÷
5
=
0
L
÷
2
L
÷
~
2
E
È
S
2
К
0
<u></u>
5
m
Ξ.
<b>S</b>
5
2
1
Е
SIC
Dĩs
: DIS
S: DIS
IS: DIS
DIS: DIS
; DIS: DIS
E; DIS: DIS
BLE; DIS: DIS
IGLE; DIS: DIS
NGLE; DIS: DIS
ANGLE; DIS: DIS
: ANGLE; DIS: DIS
<b>G: ANGLE; DIS: DIS</b>
VG: ANGLE; DIS: DIS
ANG: ANGLE; DIS: DIS
ANG: ANGLE; DIS: DIS
; ANG: ANGLE; DIS: DIS
N; ANG: ANGLE; DIS: DIS
ON; ANG: ANGLE; DIS: DIS
rion; ANG: Angle; DIS: Dis
ATION; ANG: ANGLE; DIS: DIS
AATION; ANG: ANGLE; DIS: DIS
IMATION; ANG: ANGLE; DIS: DIS
TIMATION; ANG: ANGLE; DIS: DIS
STIMATION; ANG: ANGLE; DIS: DIS
ESTIMATION; ANG: ANGLE; DIS: DIS
F ESTIMATION; ANG: ANGLE; DIS: DIS
OF ESTIMATION; ANG: ANGLE; DIS: DIS
3 OF ESTIMATION; ANG: ANGLE; DIS: DIS
PE OF ESTIMATION; ANG: ANGLE; DIS: DIS
YPE OF ESTIMATION; ANG: ANGLE; DIS: DIS
<b>LYPE OF ESTIMATION; ANG: ANGLE; DIS: DIS</b>
TYPE OF ESTIMATION; ANG: ANGLE; DIS: DIS
C: TYPE OF ESTIMATION; ANG: ANGLE; DIS: DIS
ST: TYPE OF ESTIMATION; ANG: ANGLE; DIS: DIS
3ST: TYPE OF ESTIMATION; ANG: ANGLE; DIS: DIS

			المسمر			Conc			Tratin	ation about	40.0040	
Title	FA <sup>1</sup>			00	GS		WS	OS	EST	ANG DIS	JNT	SGM
A basic study on variable-gain Kalman filter based on angle error calculated from acceleration signals for lower limb angle measurement with inertial sensors [33]	SF			>	>	>			2D	>		>
Ā Comparison of Three Neural Network Approaches for Esti- mating Joint Angles and Moments from Inertial Measurement Units [34]	ML		ML learns	>	>	>			3D <sup></sup>			1   
A fast quaternion-based orientation optimizer via virtual rota- tion for human motion tracking [116]	<u>_</u> 0Ā		     		  >			   	- <u>-</u>		     	
- A Full-State Robust Extended Kalman Filter for Orientation Tracking during Long-Duration Dynamic Tasks Using Mag- netic and Inertial Measurement Units [117]	SF	   	     	   >				   	3D		     	.>
- A Human Motion Tracking Algorithm Using Adaptive EKF Based on Markov Chain [118]	SF		     		  >			   	<u>- 3</u> <u></u> <u></u>		     	
- A method for gait analysis in a daily living environment by body-mounted instruments [35]	<u>0</u>			   			     	   	2D			I I
Ā Mēthod for Lower Back Motion Assessment Using Wearable 6D Inertial Sensors [36]	<u>0</u>	   	     				     	   	3D		     	
- A Miniature Sensor System for Precise Hand Position Moni- toring [119]	SF			   				   	<u>3</u> D <sup></sup>			-   
A new approach to accurate measurement of uniaxial joint angles based on a combination of accelerometers and gyro- scopes [37]	OA <sup>-</sup>			   			   	   				1   
Ānew quaternion-based kalman filter for human body motion tracking using the second estimator of the optimal quaternion algorithm and the joint angle constraint method with inertial and magnetic sensors [120]	SF		     	   				   	<u>3</u> <u>D</u>		   	
- A nonlinear dynamics-based estimator for functional electrical stimulation: Preliminary results from lower-leg extension experiments [38]	SF		ĒST	   			   	   	3D			1   
À novel 7 degrees of freedom model for upper limb kinematic reconstruction based on wearable sensors [121]	SF			   	  >			   	<u>-</u> 3 <u>D</u>			
Ā Novel Application of Flexible Inertial Sensors for Ambula- tory Measurement of Gait Kinematics [39]	OA		ĒST <sup>-</sup> -	   			     	   	- <u>3</u> D			,   

- 17:11		Algorit	thms			Senso	S		Estima	tion chai	acteristi	S
AULT	FA	BC	ANT	$\frac{1}{2}$	S	AS N	MS (	S	EST	ANG DI	S JNT	SGM
A novel approach to motion tracking with wearable sensors based on Probabilistic Graphical Models [122]	 ML				-			. 	3D			 
A novel complimentary filter for tracking hip angles during cycling using wireless inertial sensors and dynamic acceleration estimation [123]	SF	     	ĒŠT		-			1 1	3D			1   
A novel data glove for fingers motion capture using inertial and magnetic measurement units [124]		     	•     					1 1 1	3D		     	.>
A novel hierarchical information fusion method for three- dimensional upper limb motion estimation [125]	SF		-     					· · ·   	3D			- 
Ā Novel Kalman Filter for Human Motion Tracking With an Inertial-Based Dynamic Inclinometer [40]	SF	     	     	*	-		     	   	3D		     	- ->
A novel method for estimating knee angle using two leg- mounted gyroscopes for continuous monitoring with mobile health devices [11]	SF	     	•     		-	   	   	1 1	3D			1   
A novel method of using accelerometry for upper limb FES control [18]	<u>F</u>	     	     	   	 		     	1* * 1 1	2D		     	- ->
A novel sensor-based assessment of lower limb spasticity in children with cerebral palsy [41]	0	   	-     				   	* *   	2D+3D <sup>-</sup>			1   
A patient-centric sensory system for in-home rehabilita- tion [126]	SF	     	     	*	-			1 · · · 1 · · ·	3D			 
Ā preliminary test of measurement of joint angles and stride length with wireless inertial sensors for wearable gait evaluation system [42]	- SF	     	     		-		   	1 · · ·	2D			
Ā study of gait analysis with a smartphone for measurement of hip joint angle [12]	<u>-</u> 0A	     	     	*	-	   	     	· · ·   	2D			 
A Wearable Human Motion Tracking Device Using Micro Flow Sensor Incorporating a Micro Accelerometer [43]	SF								2D			· · · · ·
A Wearable Magnetometer-Free Motion Capture System: Inno- vative Solutions for Real-World Applications [44]	SF	   							3D			 
Accuracy Improvement on the Measurement of Human-Joint Angles [127]	<u>-</u> 0A	     	•     		-			1 · · · 1 1	3D			 
Accuracy of a custom physical activity and knee angle measure- ment sensor system for patients with neuromuscular disorders and gait abnormalities [45]	SF -		     		-		     	1 · · ·	3D			1   
Ādaptīve Gain Regulation of Sensor Fusion Algorithms for         Orientation Estimation with Magnetic and Inertial Measurement         Units [128]	SF	     						1 · · · · · · · · · · · · · · · · · · ·	3D		     	
Alignment-Free, Self-Calibrating Elbow Angles Measurement Using Inertial Sensors [13]	OA		ĒST						3D -		>	

- 1721		Algor	ithms			Sens	ors		Estima	tion chara	cteristics	I
litte	FA	BC	ANT	00	GS	AS	MS (	SC	EST	ANG DIS	JNT SG	N
Ambulatory estimation of knee-joint kinematics in anatomi-	OA			   	' '>			· · ·   	3D			
cal coordinate system using accelerometers and magnetome- ters [129]												
Ambulatory measurement and analysis of the lower limb 3D	<u></u>		>	   	$\sim$		   	1	3D <sup></sup>			1
posture using wearable sensor system [46]	     	i I I		   	  . 		   	י ד ו	     			
An adaptive complementary filter for inertial sensor based data fusion to track upper body motion [47]	SF				>	>			3D	>	>	
- Ăn auto-calibrating knee flexion-extension axis estimator using nrincinal component analysis with inertial sensors [130]	SF -		ĒST	   				· · ·   	3D			
An inertial human upper limb motion tracking method for robot	<u>-</u>	1 1 1	>-		$\leq$		   	1	3D			1
programming by demonstration [110]		   					י ו ו	י ד ו				
An inertial sensor system for measurements of tibia angle with	OA				>	>			3D	>	> >	
applications to knee valgus/varus detection [48]							   					
mation for the estimation of gait kinematics from wearable	MIL	>		>	>	>	-	<u> </u>	n	>	>	
sensors [49]												
An investigation into the accuracy of calculating upper body	SF	   	     	   	' '>		- - - - -	1 · · · 1 · ·	3D		·	
joint angles using MARG sensors [131]												
An optimized Kalman filter for the estimate of trunk orientation	SF -	   	     		- - -		 	 	2D			
from inertial sensors data during treadmill walking [50]												
Analysis of a mobile system to register the kinematic parame-	$\overline{OA}^{-1}$	   	     	   	- - -		 	     	3D			
ters in ankle, knee, and hip based in inertial sensors [51]	     	1					     		     			
Analyzing 3D knee kinematics using accelerometers, gyro-	OA	>	>		>	>	>		3D	>	>	
scopes and magnetometers [132] $= $		    			  . 		     					ī
Angle measurements during 2D and 3D movements of a rigid	SF	>		>	>	>			ZD	>	>	
outy mouter of nower mino. Comparison between meghar-based and quaternion-based methods [52]												
Artificial neural networks in motion analysis—applications of					  >		   		3D			
unsupervised and neuristic redure selection techniques [23]	<u>I</u> M	- - -	- <u>-</u>					·   	<u>-</u> <u>-</u> - <u>-</u>			I.
Using Novel Inertial Sensor-Based Technique [54]		•	learns	•	•	•			j	•	•	
Calibrated 2D angular kinematics by single-axis accelerome-	- <u>-</u>		>-	1	1 1 1	- - -	T T T		2 <u>D</u>		·	1
ters: From inverted pendulum to N-Link chain [19]		1			   		   					
Closed-chain pose estimation from wearable sensors [55]	SF 	>' >'	NO?	>¦	>' \_'	, ≻'	   		3D			,
CNN-Based Estimation of Sagittal Plane Walking and Running	ML	>		>	>	>	-		2D	>	>	
Biomechanics From Measured and Simulated Inertial Sensor Data [56]												
Comparison of angle measurements between integral-based		1 1 1	     	   	- - -	- - - -	   	1 1 1	2 <u>D</u>		· · · · · · · · · · · · · · · · · · ·	
and quaternion-based methods using inertial sensors for gait												
cvaluation [1/c]		I I I		   	   			'   				ī

- Fau		Algori	thms			Sensors		Estim	ation cl	naracter	stics
TIDE	FA	BC	ANT	OC	GS	AS MS	OS	EST	ANG	NL SIO	T SGM
Data-Driven Reconstruction of Human Locomotion Using a	ML		     	   <b>&gt;</b>				3D -	   		
The provided and the pr			- <u>M</u> L learns	   		<u>-</u>					
ics Using Shoe-Mounted IMU Sensors In Daily Living [111] - Deriving kinematic quantities from accelerometer readings for	- <u>- 40</u>							<u>3</u> <u></u>			
- The seventient of functional upper firmo motions [20] - Design and validation of an ambulatory inertial system for 3-D measurements of low back movements [59]	<u>F0</u> -			>				<u>3</u> D <sup></sup>			
Detecting absolute human knee angle and angular velocity using accelerometers and rate gyroscopes [60]	_ <u></u> 0			·>			   	$\overline{2}\overline{D}^{}$			
Development of a body joint angle measurement system using IMU sensors [61]	<u>-</u>			>			   	<u> </u>			     
Development of a wearable glove system with multiple sensors for hand kinematics assessment [133]	SF -	     	     	>  >			   	<u>3</u> <u></u> <u></u>			,     
- Development of a wearable sensor system for quantitative gait analysis [187]	O			>  >	- ->		     				
Digital inclinometer for joint angles measurements with a real- time 3D-animation [153]	- <u>-</u>	   	     	  >	   		     	<u>3</u> D <sup>-</sup>			:     
Drift-Free 3D Orientation and Displacement Estimation for Quasi-Cyclical Movements Using One Inertial Measurement Unit: Application to Running [112]	_ <u></u> 67	'     	     	  >			     	<u>3</u> <u></u> <u></u>			
- Drift-Free and Self-Aligned IMU-Based Human Gait Tracking System with Augmented Precision and Robustness [62]	SF -		     				     	<u>3</u> <u></u> <u></u>			,     
Drift-Free Foot Orientation Estimation in Running Using Wear- able IMU [63]	_0Ā^-	   	ĒST				   	3 <u>D</u>		1     	
Drift-Free Inertial Sensor-Based Joint Kinematics for Long- Term Arbitrary Movements [64]	SF -		ĒST	  >			     	3 <u></u> D <sup>-</sup>			:     
Ēffect of walking variations on complementary filter based inertial data fusion for ankle angle measurement [65]	SF -	   	ĒST <sup>-</sup>	   			     	<u>3</u> <u></u>			:     
Ēfficiency of deep neural networks for joint angle modeling in digital gait assessment [134]	ML		- <u>M</u> L - learns	<b>`</b> >							1     
Ēstimāte of lower trunk angles in pathological gaits using gyroscope data [14]	SF	'     	     	   		     	     	3D		     	
Ēstimāting Lower Body Kinematics Using a Lie Group Constrained Extended Kalman Filter and Reduced IMU Count [135]	_SF						     	<u>3</u> <u></u> <u></u>		     	
Estimating lower extremity running gait kinematics with a single accelerometer: A deep learning approach [21]	ML	>		>	1	>	>	3D	>	>	

		Algor	ithms			Senso	S	Estin	nation	charac	eristics
LILLE	FA	BC	ANT	OC	GS	AS N	AS OS	EST	ANG	DIS	JNT SGM
Ēstimāting Lower Limb Kinematics Using a Reduced Wearable Sensor Count [136]	SF							<u>3</u> <u>D</u>			
- Estimation and Observability Analysis of Human Motion on Lie Groups [66]	_SF -	   	     	>  >			     	3 <u>D</u>		   	
- Estimation of gait kinematics and kinetics from inertial sensor data using optimal control of musculoskeletal models [67]	<u>0</u>	   					     			   	
- Estimation of Gait Mechanics Based on Simulated and Mea- sured IMU Data Using an Artificial Neural Network [68]			1     		-  >			<u>3</u> <u></u>		     	
· Estimation of kinematics from inertial measurement units using a combined deep learning and optimization framework [69]			- ML - Iearns		  >			<u>3</u> <u>D</u>		   	
Ēstimātion of knee joint angle during gait cycle using inertial measurement unit sensors: a method of sensor-to-clinical bone calibration on the lower limb skeletal model [113]			     	   			     	3D		   	
- Estimation of lower limb joint angles during walking using extended kalman filtering [70]	SF	   	     		-  >		     	<u>3</u> <u></u>		     	
Ēstimātion of the continuous walking angle of knee and ankle (Talocrural joint, subtalar joint) of a lower-limb exoskeleton robot using a neural network [71]			_ ML _ learns							   	
Ēstimātion of the knee flexion-extension angle during dynamic sport motions using body-worn inertial sensors [72]	SF	   	     	  >			     			   	
- Evaluation of wearable gyroscope and accelerometer sensor (PocketIMU2) during walking and sit-to-stand motions [15]		   	     	   	  >	,     	     	<u>3</u> <u></u>		   	
Feasibility of a Wearable, Sensor-based Motion Tracking Sys- tem [16]	O	   	     	   	  >	,     	     	<u>3</u> <u>D</u>		   	
Feasibility study of inertial sensor-based joint moment estima- tion method during human movements: A test of multi-link modeling of the trunk segment [17]	_0 <u>A</u> _			   		-     	     			   	- - - -
- Gait posture estimation using wearable acceleration and gyro sensors [73]	_0 <u>A</u> _				  >		     			   	
Geometrical kinematic modeling on human motion using method of multi-sensor fusion [74]	SF			   	  >		     	<u>3</u> <u>D</u>	,     		
Grasp pose estimation in human-robot manipulation tasks using wearable motion sensors [75]		   	     	   			     	<u>3</u> D			
Hand Motion Measurement using Inertial Sensor System and Accurate Improvement by Extended Kalman Filter [76]	SF						     	<u>3</u> D			
Human Arm Motion Tracking by Inertial/Magnetic Sensors Using Unscented Kalman Filter and Relative Motion Con- straint [137]	SF -							3 <u>D</u>			
Human body model based inertial measurement of sit-to-stand motion kinematics [77]	SF	>						2D	>		

		Algori	thms			Senso	rs		Estima	tion c	haracte	ristics
liue	FA	BC	ANT	0C	GS	AS I	MS (	E E	ST	ANG	DIS JI	NT SGM
Human Motion Kinematics Assessment Using Wearable Sen- sors [138]	SF	>						   	D		>   	     
- Human motion tracking based on complementary Kalman fil- 1 ter [139]	SF	   	     	1				ו ד שו	Ū		     	
- Human pose recovery for rehabilitation using ambulatory sen- sors [78]	SF			   	.>		   	1 1 1	D -		     	
- Human pose recovery using wireless inertial measurement units [79]	SF		- - - - -	1			     	ا ا	D -			i       
· IMU-Based joint angle measurement for gait analysis [80] ]	_0 <u>A</u> _	- - - -	ĒST	   		- - ->	     	1 1 1	<u>Б</u> '	- - - 	>     	1       
- Înertia-based angle measurement unit for gait assistive de- vice [22]	OA	     	,       	 	 		     	101 101 1	<u>-</u> 0		>     	i       
- Înertial Motion Capture Using Adaptive Sensor Fusion and Joint Angle Drift Correction [140]	SF		       					ا ا	D		     	
Inertial sensing in ambulatory back load estimation [81]	<u>OA</u>							   	D			
Thertial sensor-based knee flexion/extension angle estima- tion [82]	SF		'     	, 			   	ا ا	D D		>     	i     
Joint angle estimation with accelerometers for dynamic postural analysis [23]	_0 <u>4</u> _			, 	 		     	 	n n D		>     	i     
- Joint Inertial Sensor Orientation Drift Reduction for Highly Dynamic Movements [83]	_0 <u>4</u> _		ĒST				     	ا ا	р П П		>     	1       
- Knee angle estimation based on IMU data and artificial neural networks [84]	ML		-       				<b>&gt;</b>     	ار ۱ س	D -		>     	i       
- Knee joint angle measuring portable embedded system based on inertial measurement units for gait analysis [85]	SF	   	,       	1 			     	101 101 101			>     	i       
- Lower body kinematics estimation from wearable sensors for walking and running: A deep learning approach [86]			ML				>     	ار ۱ س	р		>     	1       
- Lower Extremity Angle Measurement with Accelerome- ters—Error and Sensitivity [24]	_ <u>0</u>			   			     	10   	D -		<b>&gt;</b>     	i       
- Magneto: Joint angle analysis using an electromagnet-based sensing method [31]	<u>0</u>	   	,     	1 	 			ا ا				i       
- Magnetometer robust deep human pose regression with un- certainty prediction using sparse body worn magnetic inertial measurement units [155]			ML					ار ۱ س	ı ı ı Q		>     	i     
Monitoring of Hip and Knee Joint Angles Using a Single Iner- tial Measurement Unit during Lower Limb Rehabilitation [87]	<u>0</u> 0						     	ا ا			>     	     
- Multi-Inertial Sensor-Based Arm 3D Motion Tracking Using Elman Neural Network [114]		   	'     	 			<b>&gt;</b>   	ι΄ Ι Ιω	D D			i     
Nonlinear optimization for drift removal in estimation of gait kinematics based on accelerometers [25]	0 <u>A</u>											

Tit.		Algor	ithms			Senso	LS		Estima	tion c	haract	eristics
THE	FA	BC	ANT	OC	GS	AS	MS (	S	EST	ANG	DIS J	NT SGN
Novel approach to ambulatory assessment of human segmental orientation on a wearable sensor system [26]	OA	>		>								
- Online tracking of the lower body joint angles using IMUs for gait rehabilitation [88]	SF			   			   	1 1 1	3D			
Öptimization of Inertial Sensor-Based Motion Capturing for Magnetically Distorted Field Applications [141]	OĀ	   	   						3D		,     	
Physical-Sensor and Virtual-Sensor Based Method for Estima- tion of Lower Limb Gait Posture Using Accelerometers and Gyroscopes [89]	_0 <u>4</u> _			   			   	~ /   	3D <sup></sup>			
Pose estimation by extended Kalman filter using noise covari- ance matrices based on sensor output [142]	SF	   	,       					<`'   	<u>-</u>		,     	
Predicting Knee Joint Kinematics from Wearable Sensor Data in People with Knee Osteoarthritis and Clinical Considerations for Future Machine Learning Models [115]		 	'     	   			>   		2D <sup></sup>		'   	
Predicting lower limb joint kinematics using wearable motion sensors [90]	_ ML		   				>     		2D		,     	
Prediction of lower limb joint angles and moments during gait using artificial neural networks [91]							>       		3D		'     	
Prediction of lower limb kinetics and kinematics during walking by a single IMU on the lower back using machine learning [27]	ML	>	<b>`</b>	>					2D		>	>
Quasi-real time estimation of angular kinematics using single- axis accelerometers [92]	SF	>	>		>	>	   		2D			
Real-time estimate of body kinematics during a planar squat task using a single inertial measurement unit [93]	SF	>	>		>	>	   		2D			
Real-Time Human Motion Capture Based on Wearable Inertial Sensor Networks [152]	SF	 							3D - 1			
Reconstructing an accelerometer-based pelvis segment for three-dimensional kinematic analyses during laboratory simu- lated tasks with obstructed line-of-sight [28]	_0 <u>4</u> _	 			   		   		2D		'   	
Reconstruction of angular kinematics from wrist-worn inertial sensor data for smart home healthcare [94]	SF			     			   				,     	
- Reducing drifts in the inertial measurements of wrist and elbow positions [95]	SF			   			     		3D	   		
Research on Human Motion Monitoring Method Based on Multi-Joint Constraint Filter Model [143]	SF								3D - 1		'     	
Rhythmic EKF for pose estimation during gait [96]	SF -						     		3D	       >	  	
Rhythmic Extended Kalman Filter for Gait Rehabilitation Mo- tion Estimation and Segmentation [8]	SF 	>   	>   	> ¦	>   	>   	   		D  	> ''		

, 11:11 11:11		Algor	ithms			Sens	ors		Estim	ation c	haracte	ristics
AULT	FA	BC	ANT	0C	S	$\mathbf{AS}$	SM	OS	EST	ANG	DIS J	NT SGN
Rigid body motion capturing by means of a wearable inertial and magnetic MEMS sensor assembly-from reconstitution of the posture toward dead reckoning: An application in bio- logging [144]	SF	   	,       	•   				   	3 <u>D</u>			
Robust and Accurate Capture of Human Joint Pose Using an Inertial Sensor [145]	SF -	   	       		-			   	<u></u>		     	
- Sensorial system for obtaining the angles of the human move- ment in the coronal and sagittal anatomical planes [97]	SF		     	•     		- 	 	   	- <u>-</u>		-     	
5houlder and elbow joint angle estimation for upper limb reha- bilitation tasks using low-cost inertial and optical sensors [98]	SF			-     			   	   	$\overline{3}\overline{D}$		-     	
- Shoulder and elbow joint angle tracking with inertial sen- sors [99]	SF -			•     	-	- - 	   	   	3D		-     	,       
- Synergy-based knee angle estimation using kinematics of thigh [100]	ML		- ML				   		<u>3</u>		-	
System for measurement of joint range of motion using inertial sensors [101]	SF		'       	-     			   	   	2D		-     	'     
The head mouse - Head gaze estimation 'In-the-Wild' with low-cost inertial sensors for BMI use [146]	SF	   	,       			- - 		   	<u>-</u> 3D		     	
The manumeter: A wearable device for monitoring daily use of the wrist and fingers [32]	O	 							_ <u>3</u> D			       
The online estimation of the joint angle based on the gravity acceleration using the accelerometer and gyroscope in the wireless networks [102]	OA					>			2D	>		
The use of synthetic IMU signals in the training of deep learning models significantly improves the accuracy of joint kinematic predictions [103]			- <u>M</u> L		-	- 	   		<u>- 3</u> <u></u>		•   	'     
Three dimensional gait analysis using wearable acceleration and gyro sensors based on quaternion calculations [104]	_0 <u>4</u> _	   	-       				   	   	$\overline{3}\overline{D}$		-     	
Time coherent full-body poses estimated using only five inertial sensors: Deep versus shallow learning [147]		  >	       			- - 		   	3D			
Towards a portable human gait analysis $\&$ monitoring system [105]	0		ĒST			  >	   	   			-	
Towards kinematically constrained real time human pose esti- mation using sparse IMUs [148]			_ ML			' ' '>		   	3D		-     	'     
Training Data Selection and Optimal Sensor Placement for Deep-Learning-Based Sparse Inertial Sensor Human Posture Reconstruction [29]	ML		ML learns						<u>3</u> D		     	
Ubiquitous human upper-limb motion estimation using wear- able sensors [149]	SF -								3D		<b>-</b>       	

		Algor	ithms			Sen	sors		Estin	nation cl	haracter	istics
TIDE	FA	BC	ANT	00	GS	AS	MS	OS	EST	ANG	Nf SIQ	T SGM
Unrestrained measurement of arm motion based on a wearable	- <sup>-</sup> VO	   	   		 	$\sim$	-     	   	$2\overline{D}^{-1}$		`> - - -	   
wireless sensor network [30]												
Upper limb joint angle measurement in occupational	_0Ā_				  >	$\sim$		   	$\frac{1}{3}\overline{D}^{-1}$			-     
health [150]												
Upper limb joint angle tracking with inertial sensors [106]	SF -			     			     	   	_ <u>3</u> D _			-     
Use of multiple wearable inertial sensors in upper limb motion	_ <u>0</u> 4_			     			     	   	- <u>3</u> D -			-     
tracking [107]												
Validation of a low-cost inertial exercise tracker [108]	SF -		ĒST		  >		'     	   	- <u>3</u> D			-     
Visual and quantitative analysis of lower limb 3D gait posture	_0 <u>4</u> _			       	   			   	- <u>3</u> D -			-     
using accelerometers and magnetometers [154]												
Visual estimation of lower limb motion using physical and	_0 <u>4</u> _			       		- 		   	_ <u>3</u> D _		'> ' ' '	-     
virtual sensors [151]												
Wearable inertial sensor system towards daily human kinematic	_ML,		   			- 	'     		$\overline{2D}^{-1}$			
gait analysis: Benchmarking analysis to MVN BIOMECH [109]	$\mathbf{SF}$											

ī

ī

$\geq$
Ц
ГAВ

RELEVANT DETAILS ABOUT THE MONITORED ANATOMIC UNIT, THE NS OF STUDY IN THE SELECTED STUDIES THE VALIDATION SYSTEM, AND THE METRICS FOR EVALUATION. LG: LOWER LIMB GROUP; UG: UPPER LIMB GROUP; NS: NUMBER OF SUBJECTS FOR THE ANALYSIS; DSS: PRESENCE OF SUBJECTS WITH DISEASES; VS: VALIDATION SENSORY SYSTEM; AND M1 AND M2: METRICS PROVIDED FOR THE VALIDATION OF THE PROPOSALS

Title	Monitored anatomic unit	Subjects of study	Vali	dation approach
	LG UG	NS DSS	SN	M1 M2
Ā bāsic study on variable-gain Kalman filter based on angle error calculated from acceleration signals for lower limb angle measurement with inertial sensors [33]	Leg Feet	ιςς 	-Optical 3D	- RMSE
A Comparison of Three Neural Network Approaches for Estimating Joint Angles and Moments from Inertial Measurement Units [34]	- Ĥip/knee/ankle	<u>1</u> 16		Difference in nRMSE between methods
A fast quaternion-based orientation optimizer via virtual rotation for human motion tracking [116]		             	Optical 3D	- RMSE
A Full-State Robust Extended Kalman Filter for Ori- entation Tracking during Long-Duration Dynamic Tasks Using Magnetic and Inertial Measurement Units [117]	Leg Feet		Optical 3D	RMSE
A Human Motion Tracking Algorithm Using Adaptive EKF Based on Markov Chain [118]	Arm		Optical 3D	RMSE
A method for gait analysis in a daily living environment by body-mounted instruments [35]	Hip/knee/ankle	9	Optical 3D, Force plates	RMSE -
A Method for Lower Back Motion Assessment Using Wearable 6D Inertial Sensors [36]	Back/trunk/torso	8	Optical 3D	RMSE -
A Miniature Sensor System for Precise Hand Position Monitoring [119]		9	Optical 3D	RMSE
A new approach to accurate measurement of uniaxial joint angles based on a combination of accelerometers and gyroscopes [37]	Hip/knee/ankle	∞ ∞	Visual 2D	RMSE Corr. coe
A new quaternion-based kalman filter for human body motion tracking using the second estimator of the opti- mal quaternion algorithm and the joint angle constraint method with inertial and magnetic sensors [120]	Arm		Optical 3D	RMSE
A nonlinear dynamics-based estimator for functional elec- trical stimulation: Preliminary results from lower-leg ex- tension experiments [38]	Hip/knee/ankle	ω                 	Other	RMSE -

			-		
Title	Monitored anatomic unit	Subjects of stu NS DSS	dy Val	lidation appr ' M1	oach M7
A novel 7 degrees of freedom model for upper limb kinematic reconstruction based on wearable sensors [121]	Arm Arm joints: shoul- ders/elbow/forearm twist		Optical 3D	- RMSE -	
Ā Novel Application of Flexible Inertial Sensors for Ambulatory Measurement of Gait Kinematics [39]	- Hip/knee/ankle	3	Optical 3D	ĒRMSE	1           
Ā novel approach to motion tracking with wearable sensors based on Probabilistic Graphical Models [122]	ders/elbow/forearm	             	Optical 3D	ĒRMSE	
Ā novel complimentary filter for tracking hip angles during cycling using wireless inertial sensors and dynamic acceleration estimation [123]	- Hip/knec/ankle	             	_Optical 3D _		
A novel data glove for fingers motion capture using inertial and magnetic measurement units [124]	Hand/wrist/fingers		Vīsual 2D, Recon- struction	RMSE	
A novel hierarchical information fusion method for three- dimensional upper limb motion estimation [125]	Arm Arm joints: shoul- ders/elbow/forearm twist	4	Optical 3D	RMSE	1
A Novel Kalman Filter for Human Motion Tracking With an Inertial-Based Dynamic Inclinometer [40]	Arm	10	Optical 3D	RMSE	
A novel method for estimating knee angle using two leg-mounted gyroscopes for continuous monitoring with mobile health devices [11]	Hip/knee/ankle	5	Optical 3D	LAM	Avg. error
A novel method of using accelerometry for upper limb FES control [18]	Arm	2 2 post-stro	ke Optical 3D	MĀĒ	
A novel sensor-based assessment of lower limb spasticity in children with cerebral palsy [41]	Hip/knee/ankle	<u>33 28 pati</u> cerebral pa	ents Optical 3D	RMSE	Test-retest inter-rater reliabilities
A patient-centric sensory system for in-home rehabilita- tion [126]	Arm joints: shoul- ders/elbow/forearm twist Hand/wrist/fingers	           	Other	RMSE	1 1 1 1 1 1 1
A preliminary test of measurement of joint angles and stride length with wireless inertial sensors for wearable gait evaluation system [42]	Hip/knee/ankle Feet	ιςς   	Optical 3D	RMSE	Corr. coeff.
A study of gait analysis with a smartphone for measure- ment of hip joint angle [12]	Hip/knee/ankle	14 	Optical 3D	RMSE	

4006439

Title	Monitored anatomic unit	Subjects of study NS DSS	Vali	dation approach
Ā Wearable Human Motion Tracking Device Using Micro Flow Sensor Incorporating a Micro Accelerometer [43]				ŘMSE – – – – – – – – – – – – – – –
Ā Weārāble Magnetometer-Free Motion Capture System: Innovative Solutions for Real-World Applications [44]	ders/elbow/forearm		Optical 3D	- RMSE
Accuracy Improvement on the Measurement of Human- Joint Angles [127]	ders/elbow/forearm	. <u></u>	Optical 3D	- RMSE
Accuracy of a custom physical activity and knee angle measurement sensor system for patients with neuromus- cular disorders and gait abnormalities [45]	- Hip/knee/ankle	10	Goniometer	⁻RMSE
Ādaptīve Gain Regulation of Sensor Fusion Algorithms for Orientation Estimation with Magnetic and Inertial Measurement Units [128]	Leg Feet		Optical 3D	ĔŔMSE
- Alignment-Free, Self-Calibrating Elbow Angles Measure- ment Using Inertial Sensors [13]	ders/elbow/forearm		Optical 3D	- RMSE
Āmbulātory estimātion of knee-joint kinemātics in anatomical coordinate system using accelerometers and magnetometers [129]	Hip/knee/ankle	. <b>5</b>	Optical 3D	RMSE
- Ambulatory measurement and analysis of the lower limb 3D posture using wearable sensor system [46]	- Hip/knee/ankle		Optical 3D	- RMSE
Ān ādaptive complementary filter for inertial sensor based data fusion to track upper body motion [47]	ders/elbow/forearm		Optical 3D	- RMSE
An auto-calibrating knee flexion-extension axis estimator using principal component analysis with inertial sen- sors [130]	_Hip/knee/ankle	15	Optical 3D	RMSE
An inertial human upper limb motion tracking method for robot programming by demonstration [110]				- <u>MAE</u>
An inertial sensor system for measurements of tibia angle with applications to knee valgus/varus detection [48]	Leg	3	Optical 3D	- RMSE
An instance-based algorithm with auxiliary similarity information for the estimation of gait kinematics from wearable sensors [49]	_Hip/knee/ankle	 	Optical 3D	Absolute
An investigation into the accuracy of calculating upper body joint angles using MARG sensors [131]	Back/trunk/torso Arm joints: shoul- ders/elbow/forearm twist Hand/wrist/fingers		RBG ++ depth	RMSD

		- - - -		
Title	Monitored anatomic unit LG UG	Subjects of study NS DSS		aation approacn M1 M2
An optimized Kalman filter for the estimate of trunk orientation from inertial sensors data during treadmill walking/citeMazza2012		18	Optical 3D	RMSE
- Analysis of a mobile system to register the kinematic parameters in ankle, knee, and hip based in inertial sensors [51]	- Hip/knee/ankle	- - - - - - - - - - - - - - - - - - -	-Visual 2D	Max. error
- Analyzing 3D knee kinematics using accelerometers, gy- roscopes and magnetometers [132]	- Hip/knee/ankle		Optical 3D	Avg. error
Āngle measurements during 2D and 3D movements of a rigid body model of lower limb: Comparison between integral-based and quaternion-based methods [52]			-Optical 3D	RMSE
Artificial _ neural _ networks _ in _ motion _ analy- sis—applications of unsupervised and heuristic feature selection techniques [53]	- Hip/knee/ankle	115 <sup>-24</sup> knee <sup>-</sup> arthroplasty	-Optical 3D	ŘMŠE - Čorr. coeff.
Biomechanical Ambulatory Assessment of 3D Knee An- gle Using Novel Inertial Sensor-Based Technique [54]	- Hip/knee/ankle	5	Optical 3D	ŘMSE
Calibrated 2D angular kinematics by single-axis ac- celerometers: From inverted pendulum to N-Link chain [19]	Back/trunk/torso	1	Optical 3D	RMSE
Closed-chain pose estimation from wearable sensors [55]	Hip/knee/ankle	3	Optical 3D	Avg. error RMSE
CNN-Based Estimation of Sagittal Plane Walking and Running Biomechanics From Measured and Simulated Inertial Sensor Data [56]	Hip/knee/ankle	10	Optical 3D, Force plates	RMSE
Comparison of angle measurements between integral- based and quaternion-based methods using inertial sensors for gait evaluation [57]	Leg Feet	ι	-Optical 3D	RMSE
Data-Driven Reconstruction of Human Locomotion Using a Single Smartphone [58]			-Optical 3D	RMSE
DeepBBWAE-Net: A CNN-RNN Based Deep Super- Learner For Estimating Lower Extremity Sagittal Plane Joint Kinematics Using Shoe-Mounted IMU Sensors In Daily Living [111]	_ Hip/knee/ankle	10	Optical 3D	RMSE
Deriving kinematic quantities from accelerometer read- ings for assessment of functional upper limb motions [20]		               	Optical 3D	- <u>R</u> MSE
Design and validation of an ambulatory inertial system for 3-D measurements of low back movements [59]	Back/trunk/torso	2	Optical 3D	RMSE
Detecting absolute human knee angle and angular velocity using accelerometers and rate gyroscopes [60]	Hip/knee/ankle	2 - 1 Paraplegic	Goniometer	RMSE

4006439

	Monitored anatomic unit	Subjects of study	Vali	dation approach
	DU DI	SSG SN	SV	M1 M2
Development of a body joint angle measurement system using IMU sensors [61]	_Hip/knee/ankle		Optical 3D	Avg. error
Development of a wearable glove system with multiple sensors for hand kinematics assessment [133]		               	Goniometer	- MAE
Development of a wearable sensor system for quantitative gait analysis [46]	_Leg	10	Optical 3D	ŘMŠE – Čorr. coeff.
Digital inclinometer for joint angles measurements with a real-time 3D-animation [153]			Goniometer	Max. error
<sup>-</sup> Drift-Free and Self-Aligned IMU-Based Human Gait Tracking System with Augmented Precision and Robustness [62]	- Ĥip/knee/ankie		Optical 3D	RMSE
Drift-Free Foot Orientation Estimation in Running Using Wearable IMU [63]	Feet	26	Optical 3D	RMSE
Drift-Free 3D Orientation and Displacement Estimation for Quasi-Cyclical Movements Using One Inertial Mea- surement Unit: Application to Running [112]	Leg	14 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Optical 3D	RMSE
Drift-Free Inertial Sensor-Based Joint Kinematics for Long-Term Arbitrary Movements [64]	Hip/knee/ankle	11	Optical 3D	RMSE
Effect of walking variations on complementary filter based inertial data fusion for ankle angle measure- ment [65]	Hip/knee/ankle	10	Optical 3D	RMSE
Efficiency of deep neural networks for joint angle mod- eling in digital gait assessment [134]	Hip/knee/ankle	20	ĪMŪ	RMSE
Estimate of lower trunk angles in pathological gaits using gyroscope data [14]	Back/trunk/torso	27 13 hemiplegia, 14 Parkinson	Optical 3D	RMSE
Estimation of knee joint angle during gait cycle using inertial measurement unit sensors: a method of sensor- to-clinical bone calibration on the lower limb skeletal model [113]	Leg Hip/knee/ankle	∞ ∞	Optical 3D	RMSE
Ēstimāting Lower Body Kinematics Using a Lie Group Constrained Extended Kalman Filter and Reduced IMU Count [135]	_LegHip/knee/ankle	14	Optical 3D	- MAE
Ēstimating lower extremity running gait kinematics with a single accelerometer: A deep learning approach [21]	_Hip/knee/ankle	10	Optical 3D	RMSE
Estimating Lower Limb Kinematics Using a Reduced Wearable Sensor Count [136]	Hip/knee/ankle	6	Optical 3D	RMSE
Estimation and Observability Analysis of Human Motion on Lie Groups [66]			Optical 3D	MAE

		-		
Title	Monitored anatomic unit L.G. U.G.	Subjects of study NS DSS	VS Vali	dation approach M1
				DATOE
estimation of gatt kinematics and kinetics from metual sensor data using optimal control of musculoskeletal	Feivis Leg Hip/knee/ankle	01	Oplical 3D, Force	NINE
models [67]	4		plates	
Ēstimation of Gait Mechanics Based on Simulated and Measured IMU Data Using an Artificial Neural Net-	- Hip/knee/ankle	<u>30</u> - 24 Knee arthroplasty	Optical3D, Force	ŔMSE
work [68]		•	plates	
Estimation of kinematics from inertial measurement units	Hip/knee/ankle	1000 running-	Optical 3D	RMSE
using a combined deep learning and optimization frame- work [69]		related injury		
Estimation of lower limb joint angles during walking		3	Optical 3D	- <u>M</u> AE
using extended kalman filtering [70]	Hip/knee/ankle			
Estimation of the continuous walking angle of knee and ankle (Talocrural joint, subtalar joint) of a lower-limb	Hip/knee/ankle	1	Other	MAE
exoskeleton robot using a neural network [71]				
Estimation of the knee flexion-extension angle during dy- namic short motions using body-worn inertial sensors [72]	Leg Hin/knee/ankle	7	Optical 3D	RMSE
Evaluation of wearable gyroscope and accelerometer sen- sor (PocketIMU2) during walking and sit-to-stand mo- tions [15]	Leg Hip/knee/ankle	-	Uptical 3D	KMSE
Feasibility of a Wearable, Sensor-based Motion Tracking		8	Optical 3D	RMSE
System [16]	joints: shoul-			
	ders/elbow/forearm twist			
Feasibility study of inertial sensor-based joint moment	Back/trunk/torso	3	Optical 3D	
estimation method during human movements: A test of multi-link modeling of the trunk segment [17]				
- Gait posture estimation using wearable acceleration and gyro sensors [73]	- Hip/knee/ankle		Optical 3D	ŔMSE
- Geometrical kinematic modeling on human motion using method of multi-sensor fusion [74]	- Hip/knee/ankle	1	Other	ŘMSE
Grasp pose estimation in human-robot manipulation tasks		               	Other	Absolute
using wearable motion sensors [7,5]				errors
Hand Motion Measurement using Inertial Sensor System	Arm joints: shoul-		Optical 3D	RMSE
and Accurate Intervention by Extended Namian Fit- ter [76]	twist			
	Hand/wrist/fingers			
Human Arm Motion Tracking by Inertial/Magnetic Sen-	Arm joints: shoul-		Optical 3D	RMSE
sors Using Unscented Kalman Filter and Relative Motion	ders/elbow/forearm			
Constraint [137]	twist Hand/wrist/fingers			
	L 2			

	Monitored anatomic unit	Subjects of study	Vali	dation approach
	LG UG	NS_DSS	SN	M1M2
Human body model based inertial measurement of sit-to- stand motion kinematics [77]	Back/trunk/torso	1	Optical 3D	RMSE
Human Motion Kinematics Assessment Using Wearable Sensors [138]	- Hip/knee/ankle		Optical       -         3D, Force       -         plates       -	Ābsolute
Human motion tracking based on complementary Kalman filter [139]			Optical 3D	- <u>RMSE</u>
Human pose recovery for rehabilitation using ambulatory sensors [78]		7 Joint	Other	ĒRMSE
Human pose recovery using wireless inertial measurement units [79]	- Hip/knee/ankle	20	Optical 3D	- RMSE
- ÎMŪ-Based joint angle measurement for gait analysis [80]	- Ĥip/knee/ankle	1 - Transfemoral amputee	Optical 3D	- <u>RMSE</u>
Inertia-based angle measurement unit for gait assistive device [22]	Hip/knee/ankle		Goniometer	
Inertial Motion Capture Using Adaptive Sensor Fusion and Joint Angle Drift Correction [140]	- Hip/knee/ankle		Optical 3D	Error
· Inertial sensing in ambulatory back load estimation [81]	Back/trunk/torso	6	Optical 3D	$\tilde{\kappa}$ Error
Inertial sensor-based knee flexion/extension angle estima- tion [82]	- Hip/knee/ankle		Optical 3D	ŘMSE
Joint angle estimation with accelerometers for dynamic postural analysis [23]	- Hip/knee/ankle		Other	ŘMSE
Joint Inertial Sensor Orientation Drift Reduction for Highly Dynamic Movements [83]			Visual 2D	- <u>MAE</u>
Knee angle estimation based on IMU data and artificial neural networks [84]	Hip/knee/ankle		Goniometer	RMSE
Knee joint angle measuring portable embedded system based on inertial measurement units for gait analysis [85]	Hip/knee/ankle	12	Visual 2D	RMSE
Lower body kinematics estimation from wearable sensors for walking and running: A deep learning approach [86]	- Pelvis Hip/knee/ankle	27	Optical 3D	MĀĒ
Lower Extremity Angle Measurement with Accelerome- ters-Error and Sensitivity [24]	- Ĥip/knee/ankle		Optical 3D	- <u> </u>
Magneto: Joint angle analysis using an electromagnet- based sensing method [31]		13	Goniometer	- <u>MAE</u>
Magnetometer robust deep human pose regression with uncertainty prediction using sparse body worn magnetic inertial measurement units [155]			Other	ŘMSE

	,	, , , , , , , , , , , ,		
Title	Monitored anatomic unit LG UG	Subjects of study NS DSS	VS Vall	dation approach M1 M2
Monitoring of Hip and Knee Joint Angles Using a Single Inertial Measurement Unit during Lower Limb Rehabili- tation [87]	- Hip/knee/ankle	10	Optical 3D	RMSD
Multi-Inertial Sensor-Based Arm 3D Motion Tracking Using Elman Neural Network [114]			-Optical 3D	MAE
Nonlinear optimization for drift removal in estimation of gait kinematics based on accelerometers [25]	_Leg	10	Optical 3D	RMSE
Novel approach to ambulatory assessment of human seg- mental orientation on a wearable sensor system [26]	Leg		Optical 3D	ŘMSE
- Online tracking of the lower body joint angles using IMUs for gait rehabilitation [88]	- Hip/knee/ankle	5	Optical 3D	ĒMSĒ
Öptimization of Inertial Sensor-Based Motion Capturing for Magnetically Distorted Field Applications [141]		8 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Optical 3D	RMSE
Physical-Sensor and Virtual-Sensor Based Method for Es- timation of Lower Limb Gait Posture Using Accelerom- eters and Gyroscopes [89]	Leg	5	-Optical 3D	RMSE
Pose estimation by extended Kalman filter using noise covariance matrices based on sensor output [142]	Hip/knee/ankle	3	Optical 3D	RMSE
Predicting Knee Joint Kinematics from Wearable Sensor Data in People with Knee Osteoarthritis and Clinical Con- siderations for Future Machine Learning Models [115]	Hip/knee/ankle	17 - 17 osteoarthri- tis	Optical 3D	RMSE
Predicting lower limb joint kinematics using wearable motion sensors [90]	- Hip/knee/ankle	8	Optical 3D	. MAE
Prediction of lower limb joint angles and moments during gait using artificial neural networks [91]	Hip/knee/ankle	12	Optical 3D	RMSE
Prediction of Iower limb kinetics and kinematics during walking by a single IMU on the lower back using machine learning [27]	Leg Feet	<u> </u>	Optical 3D, Force plates	∕∞RMSE
Quasi-real time estimation of angular kinematics using single-axis accelerometers [92]	Hip/knee/ankle		Optical 3D	RMSE
Real-time estimate of body kinematics during a planar squat task using a single inertial measurement unit [93]	Hip/knee/ankle	8	Optical 3D	RMSE
Real-Time Human Motion Capture Based on Wearable Inertial Sensor Networks [152]	Hip/knee/ankle Feet	1	Optical 3D	RMSE
Reconstructing an accelerometer-based pelvis segment for three-dimensional kinematic analyses during laboratory simulated tasks with obstructed line-of-sight [28]	Pelvis		Optical 3D	RMSE

. FAT	Monitored anatomic unit	Subjects of study	Vali	dation approach
	LG UG	NS_DSS	SN	M1 M2
Reconstruction of angular kinematics from wrist-worn inertial sensor data for smart home healthcare [94]	Arm	2	IMU	MAE
Reducing drifts in the inertial measurements of wrist and elbow positions [95]		 	Optical 3D	RMSE
Research on Human Motion Monitoring Method Based on Multi-Joint Constraint Filter Model [143]			Goniometer	ŘMSE – – Max. error
Rhythmic EKF for pose estimation during gait [96]	- Hip/knee/ankle	3	Optical 3D	RMSE
Rhythmic Extended Kalman Filter for Gait Rehabilitation Motion Estimation and Segmentation [8]	- Hip/knee/ankle	5 - 2 post-stroke	Optical 3D	ŘMSE
Rigid body motion capturing by means of a wearable inertial and magnetic MEMS sensor assembly-from re- constitution of the posture toward dead reckoning: An application in bio-logging [144]	Feet	1	Visual 2D	MAE
Robust and Accurate Capture of Human Joint Pose Using	Head/neck/scapula	. 14	Optical 3D	- <b>Ē</b> MŠĒ
an Inertial Sensor [145]	Back/trunk/torso		& RGB + depth	
Sensorial system for obtaining the angles of the hu-	Back/trunk/torso	1	Other	RMSE
man movement in the coronal and sagittal anatomical planes [97]	Arm			
Shoulder and elbow joint angle estimation for upper limb rehabilitation tasks using low-cost inertial and optical		<u> </u>	Goniometer	RMSE
sensors [98]	twist		(             	
Shoulder and elbow joint angle tracking with inertial sensors [99]	Arm joints: shoul- ders/elbow/forearm twist	×	Optical 3D	RMSE
Synergy-based knee angle estimation using kinematics of thigh [100]	Hip/knee/ankle			RMSE
System for measurement of joint range of motion using inertial sensors [101]	- Hip/knee/ankle		Goniometer	- <u>M</u> ĀĒ
The head mouse - Head gaze estimation 'In-the-Wild' with low-cost inertial sensors for BMI use [146]				ŘMSE
The manumeter: A wearable device for monitoring daily use of the wrist and fingers [32]	Hand/wrist/fingers	<u> </u>	Goniometer	MAE
The online estimation of the joint angle based on the grav- ity acceleration using the accelerometer and gyroscope in the wireless networks [102]	Hip/knee/ankle		Optical 3D	RMSE

		, , , , , ,		,	
Title	Monitored anatomic unit LG	Subjects of study NS DSS	VS Vali	dation approach M1 M2	
The use of svnthetic INIT signals in the training of deep		$\frac{1}{30} - \frac{1}{17} - \frac{1}{17} - \frac{1}{17} - \frac{1}{17}$	Ontical 3D	RMSF	
learning models significantly improves the accuracy of joint kinematic predictions [103]		arthroplasty			
Three dimensional gait analysis using wearable accel- eration and gyro sensors based on quaternion calcula- tions [104]	- Hip/knee/ankle	2	-Optical 3D	- <u>RMSE</u>	-     
Time coherent full-body poses estimated using only five inertial sensors: Deep versus shallow learning [147]	 			- <u> </u>	   
Towards a portable human gait analysis monitoring sys- tem [105]	- Hip/knee/ankle	ι	Optical 3D	ŘMŠE – – – – – – – – –	,     
7owards kinematically constrained real time human pose estimation using sparse IMUs [148]		15		- MAE	,     
<sup>7</sup> Training Data Selection and Optimal Sensor Placement for Deep-Learning-Based Sparse Inertial Sensor Human Posture Reconstruction [29]			ĪMŪ	- MAE	,     
Ubiquitous human upper-limb motion estimation using wearable sensors [149]		14 1 1 1 1 1 1 1 1 1 1	Optical 3D	ŔMSE	     
Unrestrained measurement of arm motion based on a wearable wireless sensor network [30]	Arm joints: shoul- ders/elbow/forearm twist		Goniometer	MAE	
Upper limb joint angle measurement in occupational health [150]			Visual 2D	- Ţypical MĀĒ deviation	'     
Ūpper limb joint angle tracking with inertial sensors [106]			-Optical 3D	- Čorr. coeff.	,     
Ūse of multiple wearable inertial sensors in upper limb motion tracking [107]		4	-Optical 3D	RMSE	1     
· Validation of a low-cost inertial exercise tracker [108] -	Leg		Optical 3D	- RMSE	,     
Visual and quantitative analysis of lower limb 3D gait posture using accelerometers and magnetometers [154]	_Hip/knee/ankle	<b>5</b>	Optical 3D	_RMSE Corr.	coeff.
Visual estimation of lower limb motion using physical and virtual sensors [151]	- Hip/knee/ankle		Optical 3D	_RMSE Corr	coeff.
Wearable inertial sensor system towards daily human kinematic gait analysis: Benchmarking analysis to MVN BIOMECH [109]	- Hip/knee/ankle		ĪMŪ	RMSE	

of motions. In this line of work, the proposal of sparse-IMU utilization is promising to decrease the number of sensors in use, which is required for motion analysis in all environments. Moreover, the monitoring of complex joints, such as shoulders or hips, which are usually modeled as 3-DOF joints, should include all their DOF for a proper kinematic analysis.

With regard to the algorithms in use, the current trend moves from the Bayesian filters, which we consider the classical ones, to ML algorithms, especially deep learning algorithms. For the development of these novel proposals, more data with an accurate reference are required, as described in [185] and [186], in order to avoid the use of data from IMUs and simulations from the optical systems as ground truth. One of the main limitations of the BCs found in the literature is their generalization of use in wide and varied populations, where the constraints based on ROM and DOF exclude people with motor diseases. In this way, new proposals should be adaptable to the populations under study. Also, alternatives to obtain the IMU-joint vector based on inertial devices are needed in order to make suitable proposals that exploit the biomechanical relationships in out-of-the-lab environments.

Common data with inertial measurements and its reference are needed in order to obtain a fair comparison of the existent and new proposals. In that line of research, future proposals are required to be validated on a larger number of volunteers than in the current case. This also should ensure the variability of motions and not be focused on the gait.

# V. CONCLUSION

This work has reviewed the studies focused on human motion analysis based on IMUs. The date of publication of the reviewed papers is not limited, so we provide an overview of the proposals from the first study to the current date. This overview summarizes the algorithms, the combination of sensors, the anatomical units monitored, the subjects of study, and the validation approaches in the research of inertial monitoring. The review also focuses on the studies of the last decade, so we analyze the last trends in this research field. Most of the analyzed works focus on obtaining the 3-D estimation of the kinematics of lower limb joints, presenting a lack of studies of the upper half of the body. The Bayesian filters are still the most used methods, but their trend is to be applied less frequently, whereas the ML algorithms are being used now with a higher incidence. This review includes a description of the main algorithms used with their inputs and outputs for a better understanding of the existent methods. In this way, we show that, nowadays, these groups of algorithms present also differences in the selected sensors: Bayesian filters tend to use more the magnetometer and try to compensate for its limitations, but ML algorithms commonly rely only on gyroscopes and accelerometers. Both groups of algorithms present also differences in the range of accuracy, obtaining slightly lower maximum errors by using ML methods. This work also analyzes the proposed approaches for error reduction, highlighting the need for proposals suitable for all the population and IMU-joint calibration methods. Finally,

this work remarks the requirement of testing future proposals on a highly NS, which helps to create common databases that allow the comparison among the existent and new proposals.

# APPENDIX A

## TABLES OF THE DATA EXTRACTED

See Tables III and IV.

#### REFERENCES

- D. Fitzgerald et al., "Development of a wearable motion capture suit and virtual reality biofeedback system for the instruction and analysis of sports rehabilitation exercises," in *Proc. 29th Annu. Int. Conf. IEEE Eng. Med. Biol. Soc.*, Aug. 2007, pp. 4870–4874.
- [2] D. Feil-Seifer and M. J. Mataric, "Socially assistive robotics," in Proc. 9th Int. Conf. Rehabil. Robot. (ICORR), 2005, pp. 465–468.
- [3] V. Camomilla, E. Bergamini, S. Fantozzi, and G. Vannozzi, "Trends supporting the in-field use of wearable inertial sensors for sport performance evaluation: A systematic review," *Sensors*, vol. 18, no. 3, p. 873, Mar. 2018.
- [4] I. H. López-Nava and A. Muñoz-Meléndez, "Wearable inertial sensors for human motion analysis: A review," *IEEE Sensors J.*, vol. 16, no. 22, pp. 7821–7834, Nov. 2016.
- [5] J. Kubicek et al., "Recent trends, construction, and applications of smart textiles and clothing for monitoring of health activity: A comprehensive multidisciplinary review," *IEEE Rev. Biomed. Eng.*, vol. 15, pp. 36–60, 2022.
- [6] K. Kudrinko, E. Flavin, X. Zhu, and Q. Li, "Wearable sensor-based sign language recognition: A comprehensive review," *IEEE Rev. Biomed. Eng.*, vol. 14, pp. 82–97, 2021.
- [7] I. Weygers, M. Kok, M. Konings, H. Hallez, H. De Vroey, and K. Claeys, "Inertial sensor-based lower limb joint kinematics: A methodological systematic review," *Sensors*, vol. 20, no. 3, p. 673, Jan. 2020.
- [8] V. Joukov, V. Bonnet, M. Karg, G. Venture, and D. Kulić, "Rhythmic extended Kalman filter for gait rehabilitation motion estimation and segmentation," *IEEE Trans. Neural Syst. Rehabil. Eng.*, vol. 26, no. 2, pp. 407–418, Feb. 2018.
- [9] L. Adamowicz, R. Gurchiek, J. Ferri, A. Ursiny, N. Fiorentino, and R. McGinnis, "Validation of novel relative orientation and inertial sensor-to-segment algorithms for estimating 3D hip joint angles," *Sensors*, vol. 19, no. 23, p. 5143, Nov. 2019.
- [10] M. J. Page et al., "The PRISMA 2020 statement: An updated guideline for reporting systematic reviews," *BMJ*, vol. 372, p. 5, Mar. 2021.
- [11] E. Allseits, K. Kim, C. Bennett, R. Gailey, I. Gaunaurd, and V. Agrawal, "A novel method for estimating knee angle using two leg-mounted gyroscopes for continuous monitoring with mobile health devices," *Sensors*, vol. 18, no. 9, p. 2759, Aug. 2018.
- [12] A. Niijima, O. Mizuno, and T. Tanaka, "A study of gait analysis with a smartphone for measurement of hip joint angle," in *Proc. Asia–Pacific Signal Inf. Process. Assoc. Annu. Summit Conf. (APSIPA)*, Dec. 2014, pp. 1–4.
- [13] P. Müller, M.-A. Begin, T. Schauer, and T. Seel, "Alignment-free, selfcalibrating elbow angles measurement using inertial sensors," *IEEE J. Biomed. Health Informat.*, vol. 21, no. 2, pp. 312–319, Mar. 2017.
- [14] E. Grimpampi, V. Bonnet, A. Taviani, and C. Mazzà, "Estimate of lower trunk angles in pathological gaits using gyroscope data," *Gait Posture*, vol. 38, no. 3, pp. 523–527, Jul. 2013.
- [15] Q. An et al., "Evaluation of wearable gyroscope and accelerometer sensor (PocketIMU2) during walking and sit-to-stand motions," in *Proc. 21st IEEE Int. Symp. Robot Hum. Interact. Commun. (IEEE RO-MAN)*, Sep. 2012, pp. 731–736.
- [16] F. F. Mohammadzadeh, S. Liu, K. A. Bond, and C. S. Nam, "Feasibility of a wearable, sensor-based motion tracking system," *Proc. Manuf.*, vol. 3, pp. 192–199, Jan. 2015.
- [17] T. Watanabe and J. Kodama, "Feasibility study of inertial sensor-based joint moment estimation method during human movements—A test of multi-link modeling of the trunk segment," in *Proc. BIOSIGNALS* 9th Int. Joint Conf. Biomed. Eng. Syst. Technol. (BIOSTEC), 2016, pp. 248–255.
- [18] M. Sun et al., "A novel method of using accelerometry for upper limb FES control," *Med. Eng. Phys.*, vol. 38, no. 11, pp. 1244–1250, Nov. 2016.

- [19] F. Bagalá, V. L. Fuschillo, L. Chiari, and A. Cappello, "Calibrated 2D angular kinematics by single-axis accelerometers: From inverted pendulum to N-link chain," *IEEE Sensors J.*, vol. 12, no. 3, pp. 479–486, Mar. 2012.
- [20] D. Laidig and T. Seel, "Deriving kinematic quantities from accelerometer readings for assessment of functional upper limb motions," *Current Directions Biomed. Eng.*, vol. 3, no. 2, pp. 573–576, Sep. 2017.
- [21] M. Gholami, C. Napier, and C. Menon, "Estimating lower extremity running gait kinematics with a single accelerometer: A deep learning approach," *Sensors*, vol. 20, no. 10, p. 2939, May 2020.
- [22] R. Das, N. Kumar, and A. Kumar, "Inertia-based angle measurement unit for gait assistive device," *Int. J. Med. Eng. Informat.*, vol. 6, no. 3, pp. 266–273, 2014.
- [23] J. Ma, H. Kharboutly, A. Benali, F. Benamar, and M. Bouzit, "Joint angle estimation with accelerometers for dynamic postural analysis," *J. Biomech.*, vol. 48, no. 13, pp. 3616–3624, Oct. 2015.
- [24] A. T. M. Willemsen, C. Frigo, and H. B. K. Boom, "Lower extremity angle measurement with accelerometers-error and sensitivity analysis," *IEEE Trans. Biomed. Eng.*, vol. 38, no. 12, pp. 1186–1193, Dec. 1991.
- [25] M. D. Djurić-Jovičić, N. S. Jovičić, D. B. Popović, and A. R. Djordjević, "Nonlinear optimization for drift removal in estimation of gait kinematics based on accelerometers," *J. Biomech.*, vol. 45, no. 16, pp. 2849–2854, Nov. 2012.
- [26] K. Liu, T. Liu, K. Shibata, Y. Inoue, and R. Zheng, "Novel approach to ambulatory assessment of human segmental orientation on a wearable sensor system," J. Biomech., vol. 42, no. 16, pp. 2747–2752, Dec. 2009.
- [27] H. Lim, B. Kim, and S. Park, "Prediction of lower limb kinetics and kinematics during walking by a single IMU on the lower back using machine learning," *Sensors*, vol. 20, no. 1, p. 130, Dec. 2019.
- [28] J. D. Zehr, L. M. Tennant, J. M. Buchman-Pearle, and J. P. Callaghan, "Reconstructing an accelerometer-based pelvis segment for threedimensional kinematic analyses during laboratory simulated tasks with obstructed line-of-sight," *J. Biomech.*, vol. 123, Jun. 2021, Art. no. 110512.
- [29] Z. Zheng, H. Ma, W. Yan, H. Liu, and Z. Yang, "Training data selection and optimal sensor placement for deep-learning-based sparse inertial sensor human posture reconstruction," *Entropy*, vol. 23, no. 5, p. 588, May 2021.
- [30] G. X. Lee, K. S. Low, and T. Taher, "Unrestrained measurement of arm motion based on a wearable wireless sensor network," *IEEE Trans. Instrum. Meas.*, vol. 59, no. 5, pp. 1309–1317, May 2010.
- [31] A. Watson, A. Lyubovsky, K. Koltermann, and G. Zhou, "Magneto: Joint angle analysis using an electromagnet-based sensing method," in *Proc. 20th Int. Conf. Inf. Process. Sensor Netw.*, May 2021, pp. 1–14.
- [32] N. Friedman, J. B. Rowe, D. J. Reinkensmeyer, and M. Bachman, "The manumeter: A wearable device for monitoring daily use of the wrist and fingers," *IEEE J. Biomed. Health Informat.*, vol. 18, no. 6, pp. 2168–2194, Nov. 2014.
- [33] Y. Teruyama and T. Watanabe, "A basic study on variable-gain Kalman filter based on angle error calculated from acceleration signals for lower limb angle measurement with inertial sensors," in *Proc. 35th Annu. Int. Conf. IEEE Eng. Med. Biol. Soc. (EMBC)*, Jul. 2013, pp. 3423–3426.
- [34] M. Mundt, W. R. Johnson, W. Potthast, B. Markert, A. Mian, and J. Alderson, "A comparison of three neural network approaches for estimating joint angles and moments from inertial measurement units," *Sensors*, vol. 21, no. 13, p. 4535, Jul. 2021.
- [35] Y. Ohtaki, K. Sagawa, and H. Inooka, "A method for gait analysis in a daily living environment by body-mounted instruments," *JSME Int. J. C, Mech. Syst., Mach. Elements Manuf.*, vol. 44, no. 4, pp. 1125–1132, 2001.
- [36] M. Molnar, M. Kok, T. Engel, H. Kaplick, F. Mayer, and T. Seel, "A method for lower back motion assessment using wearable 6D inertial sensors," in *Proc. 21st Int. Conf. Inf. Fusion (FUSION)*, Jul. 2018, pp. 799–806.
- [37] H. Dejnabadi, B. M. Jolles, and K. Aminian, "A new approach to accurate measurement of uniaxial joint angles based on a combination of accelerometers and gyroscopes," *IEEE Trans. Biomed. Eng.*, vol. 52, no. 8, pp. 1478–1484, Aug. 2005.
- [38] M. Allen, Q. Zhong, N. Kirsch, A. Dani, W. W. Clark, and N. Sharma, "A nonlinear dynamics-based estimator for functional electrical stimulation: Preliminary results from lower-leg extension experiments," *IEEE Trans. Neural Syst. Rehabil. Eng.*, vol. 25, no. 12, pp. 2365–2374, Dec. 2017.

- [39] W. Yin, C. Reddy, Y. Zhou, and X. Zhang, "A novel application of flexible inertial sensors for ambulatory measurement of gait kinematics," *IEEE Trans. Human-Mach. Syst.*, vol. 51, no. 4, pp. 346–354, Aug. 2021.
- [40] G. Ligorio and A. M. Sabatini, "A novel Kalman filter for human motion tracking with an inertial-based dynamic inclinometer," *IEEE Trans. Biomed. Eng.*, vol. 62, no. 8, pp. 2033–2043, Aug. 2015.
- [41] S. Choi, Y. B. Shin, S.-Y. Kim, and J. Kim, "A novel sensor-based assessment of lower limb spasticity in children with cerebral palsy," *J. Neuroeng. Rehabil.*, vol. 15, no. 1, Dec. 2018, Art. no. 45.
- [42] T. Watanabe, H. Saito, E. Koike, and K. Nitta, "A preliminary test of measurement of joint angles and stride length with wireless inertial sensors for wearable gait evaluation system," *Comput. Intell. Neurosci.*, vol. 2011, pp. 1–12, Jan. 2011.
- [43] S. Q. Liu, J. C. Zhang, and R. Zhu, "A wearable human motion tracking device using micro flow sensor incorporating a micro accelerometer," *IEEE Trans. Biomed. Eng.*, vol. 67, no. 4, pp. 940–948, Apr. 2020.
- [44] G. Ligorio et al., "A wearable magnetometer-free motion capture system: Innovative solutions for real-world applications," *IEEE Sensors J.*, vol. 20, no. 15, pp. 8844–8857, Aug. 2020.
- [45] F. Feldhege et al., "Accuracy of a custom physical activity and knee angle measurement sensor system for patients with neuromuscular disorders and gait abnormalities," *Sensors*, vol. 15, no. 5, pp. 10734–10752, May 2015.
- [46] K. Liu, T. Liu, K. Shibata, and Y. Inoue, "Ambulatory measurement and analysis of the lower limb 3D posture using wearable sensor system," in *Proc. Int. Conf. Mechatronics Autom. (ICMA)*, Aug. 2009, pp. 3065–3069.
- [47] M. S. Karunarathne, S. W. Ekanayake, and P. N. Pathirana, "An adaptive complementary filter for inertial sensor based data fusion to track upper body motion," in *Proc. 7th Int. Conf. Inf. Autom. Sustainability (ICIAfS)*, Dec. 2014, pp. 1–5.
- [48] W. Hu, E. Charry, M. Umer, A. Ronchi, and S. Taylor, "An inertial sensor system for measurements of tibia angle with applications to knee valgus/varus detection," in *Proc. IEEE 9th Int. Conf. Intell. Sensors, Sensor Netw. Inf. Process. (ISSNIP)*, Apr. 2014, pp. 1–6.
- [49] J. Y. Goulermas et al., "An instance-based algorithm with auxiliary similarity information for the estimation of gait kinematics from wearable sensors," *IEEE Trans. Neural Netw.*, vol. 19, no. 9, pp. 1574–1582, Sep. 2008.
- [50] C. Mazzà, M. Donati, J. McCamley, P. Picerno, and A. Cappozzo, "An optimized Kalman filter for the estimate of trunk orientation from inertial sensors data during treadmill walking," *Gait Posture*, vol. 35, no. 1, pp. 138–142, Jan. 2012.
- [51] V. H. Flores-Morales, B. G. Contreras-Bermeo, F. L. Bueno-Palomeque, and L. J. Serpa-Andrade, "Analysis of a mobile system to register the kinematic parameters in ankle, knee, and hip based in inertial sensors," in *Proc. 4th Int. Congr. Sport Sci. Res. Technol. Support (icSPORTS)*, 2016, pp. 29–33.
- [52] T. Watanabe and K. Ohashi, "Angle measurements during 2D and 3D movements of a rigid body model of lower limb: Comparison between integral-based and quaternion-based methods," in *Proc. 7th Int. Conf. Bio-Inspired Syst. Signal Process. (BIOSIGNALS), 7th Int. Joint Conf. Biomed. Eng. Syst. Technol. (BIOSTEC),* 2014, pp. 35–44.
- [53] M. Mundt, A. Koeppe, F. Bamer, S. David, and B. Markert, "Artificial neural networks in motion analysis—Applications of unsupervised and heuristic feature selection techniques," *Sensors*, vol. 20, no. 16, pp. 1–15, 2020.
- [54] L. K. Tham, N. A. A. Osman, M. A. Kouzbary, and K. Aminian, "Biomechanical ambulatory assessment of 3D knee angle using novel inertial sensor-based technique," *IEEE Access*, vol. 9, pp. 36559–36570, 2021.
- [55] V. Joukov, J. F.-S. Lin, and D. Kulic, "Closed-chain pose estimation from wearable sensors," in *Proc. IEEE-RAS 19th Int. Conf. Hum. Robots (Humanoids)*, Oct. 2019, pp. 594–600.
- [56] E. Dorschky, M. Nitschke, C. F. Martindale, A. J. van den Bogert, A. D. Koelewijn, and B. M. Eskofier, "CNN-based estimation of sagittal plane walking and running biomechanics from measured and simulated inertial sensor data," *Frontiers Bioeng. Biotechnol.*, vol. 8, p. 604, Jun. 2020.
- [57] T. Watanabe, Y. Teruyama, and K. Ohashi, "Comparison of angle measurements between integral-based and quaternion-based methods using inertial sensors for gait evaluation," in *Proc. Int. Joint Conf. Biomed. Eng. Syst. Technol.*, in Communications in Computer and Information Science, vol. 511, 2015, pp. 274–288.

- [58] H. Eom, B. Choi, and J. Noh, "Data-driven reconstruction of human locomotion using a single smartphone," *Comput. Graph. Forum*, vol. 33, no. 7, pp. 11–19, Oct. 2014.
- [59] E. Charry, M. Umer, and S. Taylor, "Design and validation of an ambulatory inertial system for 3-D measurements of low back movements," in *Proc. 7th Int. Conf. Intell. Sensors, Sensor Netw. Inf. Process. (ISSNIP)*, Dec. 2011, pp. 58–63.
- [60] R. Williamson and B. J. Andrews, "Detecting absolute human knee angle and angular velocity using accelerometers and rate gyroscopes," *Med. Biol. Eng. Comput.*, vol. 39, no. 3, pp. 294–302, May 2001.
- [61] S. Bakhshi, M. H. Mahoor, and B. S. Davidson, "Development of a body joint angle measurement system using IMU sensors," in *Proc. Annu. Int. Conf. IEEE Eng. Med. Biol. Soc. (EMBS)*, Aug. 2011, pp. 6923–6926.
- [62] Y. Chen, C. Fu, W. S. W. Leung, and L. Shi, "Drift-free and self-aligned IMU-based human gait tracking system with augmented precision and robustness," *IEEE Robot. Autom. Lett.*, vol. 5, no. 3, pp. 4671–4678, Jul. 2020.
- [63] M. Falbriard, F. Meyer, B. Mariani, G. P. Millet, and K. Aminian, "Drift-free foot orientation estimation in running using wearable IMU," *Frontiers Bioeng. Biotechnol.*, vol. 8, p. 65, Feb. 2020.
- [64] I. Weygers et al., "Drift-free inertial sensor-based joint kinematics for long-term arbitrary movements," *IEEE Sensors J.*, vol. 20, no. 14, pp. 7969–7979, Jul. 2020.
- [65] L. Meng, B. Li, C. Childs, A. Buis, F. He, and D. Ming, "Effect of walking variations on complementary filter based inertial data fusion for ankle angle measurement," in *Proc. IEEE Int. Conf. Comput. Intell. Virtual Environ. Meas. Syst. Appl. (CIVEMSA)*, Jun. 2019, pp. 1–5.
- [66] V. Joukov, J. Cesic, K. Westermann, I. Markovic, I. Petrovic, and D. Kulic, "Estimation and observability analysis of human motion on lie groups," *IEEE Trans. Cybern.*, vol. 50, no. 3, pp. 1321–1332, Mar. 2020.
- [67] E. Dorschky, M. Nitschke, A.-K. Seifer, A. J. van den Bogert, and B. M. Eskofier, "Estimation of gait kinematics and kinetics from inertial sensor data using optimal control of musculoskeletal models," *J. Biomech.*, vol. 95, Oct. 2019, Art. no. 109278.
- [68] M. Mundt et al., "Estimation of gait mechanics based on simulated and measured IMU data using an artificial neural network," *Frontiers Bioeng. Biotechnol.*, vol. 8, p. 41, Feb. 2020.
- [69] E. Rapp, S. Shin, W. Thomsen, R. Ferber, and E. Halilaj, "Estimation of kinematics from inertial measurement units using a combined deep learning and optimization framework," *J. Biomech.*, vol. 116, Feb. 2021, Art. no. 110229.
- [70] D. Young, S. D'Orey, R. Opperman, C. Hainley, and D. J. Newman, "Estimation of lower limb joint angles during walking using extended Kalman filtering," in *Proc. 6th World Congr. Biomech.*, in IFMBE Proceedings, vol. 31, 2010, pp. 1319–1322.
- [71] T. Lee, I. Kim, and S.-H. Lee, "Estimation of the continuous walking angle of knee and ankle (talocrural joint, subtalar joint) of a lowerlimb exoskeleton robot using a neural network," *Sensors*, vol. 21, no. 8, p. 2807, Apr. 2021.
- [72] C. Jakob et al., "Estimation of the knee flexion-extension angle during dynamic sport motions using body-worn inertial sensors," in *Proc. 8th Int. Conf. Body Area Netw.*, 2013, pp. 289–295.
- [73] R. Takeda, S. Tadano, A. Natorigawa, M. Todoh, and S. Yoshinari, "Gait posture estimation using wearable acceleration and gyro sensors," *J. Biomech.*, vol. 42, no. 15, pp. 2486–2494, Nov. 2009.
- [74] C. Xu, J. He, X. Zhang, C. Yao, and P.-H. Tseng, "Geometrical kinematic modeling on human motion using method of multi-sensor fusion," *Inf. Fusion*, vol. 41, pp. 243–254, May 2018.
- [75] D. Cehajic, S. Erhart, and S. Hirche, "Grasp pose estimation in human-robot manipulation tasks using wearable motion sensors," in *Proc. IEEE/RSJ Int. Conf. Intell. Robots Syst. (IROS)*, Sep. 2015, pp. 1031–1036.
- [76] K. Kitano, A. Ito, and N. Tsujiuchi, "Hand motion measurement using inertial sensor system and accurate improvement by extended Kalman filter," in *Proc. 41st Annu. Int. Conf. IEEE Eng. Med. Biol. Soc.* (*EMBC*), Jul. 2019, pp. 6405–6408.
- [77] J. Musić, R. Kamnik, V. Zanchi, and M. Munih, "Human body model based inertial measurement of sit-to-stand motion kinematics," WSEAS Trans. Syst., vol. 7, no. 3, pp. 156–164, 2008.
- [78] J. F.-S. Lin and D. Kulic, "Human pose recovery for rehabilitation using ambulatory sensors," in *Proc. 35th Annu. Int. Conf. IEEE Eng. Med. Biol. Soc. (EMBC)*, Jul. 2013, pp. 4799–4802.

- [79] J. F. S. Lin and D. Kulić, "Human pose recovery using wireless inertial measurement units," *Physiol. Meas.*, vol. 33, no. 12, pp. 2099–2115, Dec. 2012.
- [80] T. Seel, J. Raisch, and T. Schauer, "IMU-based joint angle measurement for gait analysis," *Sensors*, vol. 14, no. 4, pp. 6891–6909, Apr. 2014.
- [81] C. T. M. Baten, P. Oosterhoff, I. Kingma, P. H. Veltink, and H. J. Hermens, "Inertial sensing in ambulatory back load estimation," in *Proc. 18th Annu. Int. Conf. IEEE Eng. Med. Biol. Soc.*, 1996, pp. 497–498.
- [82] G. Cooper et al., "Inertial sensor-based knee flexion/extension angle estimation," J. Biomech., vol. 42, no. 16, pp. 2678–2685, Dec. 2009.
- [83] B. Fasel, J. Sporri, J. Chardonnens, J. Kroll, E. Müller, and K. Aminian, "Joint inertial sensor orientation drift reduction for highly dynamic movements," *IEEE J. Biomed. Health Informat.*, vol. 22, no. 1, pp. 77–86, Jan. 2018.
- [84] C. L. Bennett, C. Odom, and M. Ben-Asher, "Knee angle estimation based on IMU data and artificial neural networks," in *Proc. 29th Southern Biomed. Eng. Conf.*, May 2013, pp. 111–112.
- [85] D. Mayorca-Torres, J. C. Caicedo-Eraso, and D. H. Peluffo-Ordóñez, "Knee joint angle measuring portable embedded system based on inertial measurement units for gait analysis," *Int. J. Adv. Sci., Eng. Inf. Technol.*, vol. 10, no. 2, pp. 430–437, 2020.
- [86] V. Hernandez, D. Dadkhah, V. Babakeshizadeh, and D. Kulić, "Lower body kinematics estimation from wearable sensors for walking and running: A deep learning approach," *Gait Posture*, vol. 83, pp. 185–193, Jan. 2021.
- [87] V. Bonnet, V. Joukov, D. Kulić, P. Fraisse, N. Ramdani, and G. Venture, "Monitoring of hip and knee joint angles using a single inertial measurement unit during lower limb rehabilitation," *IEEE Sensors J.*, vol. 16, no. 6, pp. 1557–1564, Mar. 2016.
- [88] V. Joukov, M. Karg, and D. Kulic, "Online tracking of the lower body joint angles using IMUs for gait rehabilitation," in *Proc. 36th Annu. Int. Conf. IEEE Eng. Med. Biol. Soc. (EMBC)*, Aug. 2014, pp. 2310–2313.
- [89] K. Liu, Y. Inoue, and K. Shibata, "Physical-sensor and virtualsensor based method for estimation of lower limb gait posture using accelerometers and gyroscopes," *J. Biomech. Sci. Eng.*, vol. 5, no. 4, pp. 472–483, 2010.
- [90] A. Findlow, J. Y. Goulermas, C. Nester, D. Howard, and L. P. J. Kenney, "Predicting lower limb joint kinematics using wearable motion sensors," *Gait Posture*, vol. 28, no. 1, pp. 120–126, Jul. 2008.
- [91] M. Mundt et al., "Prediction of lower limb joint angles and moments during gait using artificial neural networks," *Med. Biol. Eng. Comput.*, vol. 58, no. 1, pp. 211–225, Jan. 2020.
- [92] A. Caroselli, F. Bagalà, and A. Cappello, "Quasi-real time estimation of angular kinematics using single-axis accelerometers," *Sensors*, vol. 13, no. 1, pp. 918–937, Jan. 2013.
- [93] V. Bonnet, C. Mazza, P. Fraisse, and A. Cappozzo, "Real-time estimate of body kinematics during a planar squat task using a single inertial measurement unit," *IEEE Trans. Biomed. Eng.*, vol. 60, no. 7, pp. 1920–1926, Jul. 2013.
- [94] E. Villeneuve, W. Harwin, W. Holderbaum, B. Janko, and R. S. Sherratt, "Reconstruction of angular kinematics from wrist-worn inertial sensor data for smart home healthcare," *IEEE Access*, vol. 5, pp. 2351–2363, 2017.
- [95] H. Zhou and H. Hu, "Reducing drifts in the inertial measurements of wrist and elbow positions," *IEEE Trans. Instrum. Meas.*, vol. 59, no. 3, pp. 575–585, Mar. 2010.
- [96] V. Joukov, V. Bonnet, M. Karg, G. Venture, and D. Kulić, "Rhythmic EKF for pose estimation during gait," in *Proc. IEEE-RAS 15th Int. Conf. Hum. Robots (Humanoids)*, Nov. 2015, pp. 1167–1172.
- [97] D. Alvarado, L. Corona, S. Muñoz, and J. Aquino, "Sensorial system for obtaining the angles of the human movement in the coronal and sagittal anatomical planes," in *Proc. Mexican Int. Conf. Artif. Intell.*, in Lecture Notes in Computer Science: Including Subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics, vol. 10061, 2017, pp. 535–547.
- [98] A. Alizadegan and S. Behzadipour, "Shoulder and elbow joint angle estimation for upper limb rehabilitation tasks using low-cost inertial and optical sensors," J. Mech. Med. Biol., vol. 17, no. 2, Mar. 2017, Art. no. 1750031.
- [99] M. El-Gohary and J. McNames, "Shoulder and elbow joint angle tracking with inertial sensors," *IEEE Trans. Biomed. Eng.*, vol. 59, no. 9, pp. 2635–2641, Sep. 2012.

- [100] F.-Y. Liang, F. Gao, and W.-H. Liao, "Synergy-based knee angle estimation using kinematics of thigh," *Gait Posture*, vol. 89, pp. 25–30, Sep. 2021.
- [101] B. S. M. Sharma, S. Vidhya, and N. Kumar, "System for measurement of joint range of motion using inertial sensors," *Biomed. Res.*, vol. 28, no. 8, pp. 3699–3704, 2017.
- [102] Z. Ding, C. Yang, J. Ma, J. Wei, and F. Jiang, "The online estimation of the joint angle based on the gravity acceleration using the accelerometer and gyroscope in the wireless networks," *Multimedia Tools Appl.*, vol. 79, nos. 23–24, pp. 16265–16279, Jun. 2020.
- [103] M. S. Renani, A. M. Eustace, C. A. Myers, and C. W. Clary, "The use of synthetic IMU signals in the training of deep learning models significantly improves the accuracy of joint kinematic predictions," *Sensors*, vol. 21, no. 17, p. 5876, Aug. 2021.
- [104] S. Tadano, R. Takeda, and H. Miyagawa, "Three dimensional gait analysis using wearable acceleration and gyro sensors based on quaternion calculations," *Sensors*, vol. 13, no. 7, pp. 9321–9343, Jul. 2013.
- [105] S. Kumar, K. Gopinath, L. Rocchi, P. T. Sukumar, S. Kulkarni, and J. Sampath, "Towards a portable human gait analysis & monitoring system," in *Proc. Int. Conf. Signals Syst. (ICSigSys)*, May 2018, pp. 174–180.
- [106] M. El-Gohary, L. Holmstrom, J. Huisinga, E. King, J. McNames, and F. Horak, "Upper limb joint angle tracking with inertial sensors," in *Proc. Annu. Int. Conf. IEEE Eng. Med. Biol. Soc. (EMBS)*, Aug. 2011, pp. 5629–5632.
- [107] H. Zhou, T. Stone, H. Hu, and N. Harris, "Use of multiple wearable inertial sensors in upper limb motion tracking," *Med. Eng. Phys.*, vol. 30, no. 1, pp. 123–133, Jan. 2008.
- [108] S. Salehi and D. Stricker, "Validation of a low-cost inertial exercise tracker," in *Proc. 9th Int. Conf. Sensor Netw.*, 2020, pp. 97–104.
- [109] J. Figueiredo, S. P. Carvalho, J. P. Vilas-Boas, L. M. Gonçalves, J. C. Moreno, and C. P. Santos, "Wearable inertial sensor system towards daily human kinematic gait analysis: Benchmarking analysis to MVN BIOMECH," *Sensors*, vol. 20, no. 8, p. 2185, Apr. 2020.
- [110] R. Pellois and O. Brüls, "An inertial human upper limb motion tracking method for robot programming by demonstration," *Robot. Auto. Syst.*, vol. 156, Oct. 2022, Art. no. 104201.
- [111] M. S. B. Hossain, J. Dranetz, H. Choi, and Z. Guo, "DeepBBWAE-Net: A CNN-RNN based deep superlearner for estimating lower extremity sagittal plane joint kinematics using shoe-mounted IMU sensors in daily living," *IEEE J. Biomed. Health Informat.*, vol. 26, no. 8, pp. 3906–3917, Aug. 2022.
- [112] M. A. Zandbergen, J. Reenalda, R. P. van Middelaar, R. I. Ferla, J. H. Buurke, and P. H. Veltink, "Drift-free 3D orientation and displacement estimation for quasi-cyclical movements using one inertial measurement unit: Application to running," *Sensors*, vol. 22, no. 3, p. 956, Jan. 2022.
- [113] S. Wang et al., "Estimation of knee joint angle during gait cycle using inertial measurement unit sensors: A method of sensor-to-clinical bone calibration on the lower limb skeletal model," *J. Biomech. Sci. Eng.*, vol. 17, no. 1, pp. 1–15, 2022.
- [114] H. Yang et al., "Multi-inertial sensor-based arm 3D motion tracking using Elman neural network," J. Sensors, vol. 2022, pp. 1–11, May 2022.
- [115] J.-S. Tan et al., "Predicting knee joint kinematics from wearable sensor data in people with knee osteoarthritis and clinical considerations for future machine learning models," *Sensors*, vol. 22, no. 2, p. 446, Jan. 2022.
- [116] J. K. Lee and E. J. Park, "A fast quaternion-based orientation optimizer via virtual rotation for human motion tracking," *IEEE Trans. Biomed. Eng.*, vol. 56, no. 5, pp. 1574–1582, May 2009.
- [117] M. Nazarahari and H. Rouhani, "A full-state robust extended Kalman filter for orientation tracking during long-duration dynamic tasks using magnetic and inertial measurement units," *IEEE Trans. Neural Syst. Rehabil. Eng.*, vol. 29, pp. 1280–1289, 2021.
- [118] C. W. Kang, H. J. Kim, and C. G. Park, "A human motion tracking algorithm using adaptive EKF based on Markov chain," *IEEE Sensors J.*, vol. 16, no. 24, pp. 8953–8962, Dec. 2016.
- [119] R. Abbasi-Kesbi and A. Nikfarjam, "A miniature sensor system for precise hand position monitoring," *IEEE Sensors J.*, vol. 18, no. 6, pp. 2577–2584, Mar. 2018.

- [120] Y. Duan, X. Zhang, and Z. Li, "A new quaternion-based Kalman filter for human body motion tracking using the second estimator of the optimal quaternion algorithm and the joint angle constraint method with inertial and magnetic sensors," *Sensors*, vol. 20, no. 21, pp. 1–19, 2020.
- [121] L. Peppoloni, A. Filippeschi, E. Ruffaldi, and C. A. Avizzano, "A novel 7 degrees of freedom model for upper limb kinematic reconstruction based on wearable sensors," in *Proc. IEEE 11th Int. Symp. Intell. Syst. Informat. (SISY)*, Sep. 2013, pp. 105–110.
- [122] E. Ruffaldi, L. Peppoloni, A. Filippeschi, and C. A. Avizzano, "A novel approach to motion tracking with wearable sensors based on probabilistic graphical models," in *Proc. IEEE Int. Conf. Robot. Autom. (ICRA)*, May 2014, pp. 1247–1252.
- [123] J. Cockcroft, J. H. Müller, and C. Scheffer, "A novel complimentary filter for tracking hip angles during cycling using wireless inertial sensors and dynamic acceleration estimation," *IEEE Sensors J.*, vol. 14, no. 8, pp. 2864–2871, Aug. 2014.
- [124] B. Fang, F. Sun, H. Liu, and D. Guo, "A novel data glove for fingers motion capture using inertial and magnetic measurement units," in *Proc. IEEE Int. Conf. Robot. Biomimetics (ROBIO)*, Dec. 2016, pp. 2099–2104.
- [125] Z.-Q. Zhang and J.-K. Wu, "A novel hierarchical information fusion method for three-dimensional upper limb motion estimation," *IEEE Trans. Instrum. Meas.*, vol. 60, no. 11, pp. 3709–3719, Nov. 2011.
- [126] R. Abbasi-Kesbi, A. Nikfarjam, and H. Memarzadeh-Tehran, "A patient-centric sensory system for in-home rehabilitation," *IEEE Sensors J.*, vol. 17, no. 2, pp. 524–533, Jan. 2017.
- [127] D. Meng, T. Shoepe, and G. Vejarano, "Accuracy improvement on the measurement of human-joint angles," *IEEE J. Biomed. Health Informat.*, vol. 20, no. 2, pp. 498–507, Mar. 2016.
- [128] M. Nazarahari and H. Rouhani, "Adaptive gain regulation of sensor fusion algorithms for orientation estimation with magnetic and inertial measurement units," *IEEE Trans. Instrum. Meas.*, vol. 70, pp. 1–13, 2021.
- [129] L. Kun, Y. Inoue, K. Shibata, and C. Enguo, "Ambulatory estimation of knee-joint kinematics in anatomical coordinate system using accelerometers and magnetometers," *IEEE Trans. Biomed. Eng.*, vol. 58, no. 2, pp. 435–442, Feb. 2011.
- [130] T. McGrath, R. Fineman, and L. Stirling, "An auto-calibrating knee flexion-extension axis estimator using principal component analysis with inertial sensors," *Sensors*, vol. 18, no. 6, p. 1882, Jun. 2018.
- [131] E. Mazomenos, A. Cranny, D. Biswas, N. Harris, and K. Maharatna, "An investigation into the accuracy of calculating upper body joint angles using MARG sensors," *Proc. Eng.*, vol. 87, pp. 1330–1333, Jan. 2014.
- [132] K. Kawano, S. Kobashi, M. Yagi, K. Kondo, S. Yoshiya, and Y. Hata, "Analyzing 3D knee kinematics using accelerometers, gyroscopes and magnetometers," in *Proc. IEEE Int. Conf. Syst. Syst. Eng. (SOSE)*, Apr. 2007, pp. 1–6.
- [133] F. Fei et al., "Development of a wearable glove system with multiple sensors for hand kinematics assessment," *Micromachines*, vol. 12, no. 4, p. 362, Mar. 2021.
- [134] J. C. Alcaraz, S. Moghaddamnia, and J. Peissig, "Efficiency of deep neural networks for joint angle modeling in digital gait assessment," *EURASIP J. Adv. Signal Process.*, vol. 2021, no. 1, pp. 1–20, Dec. 2021.
- [135] L. W. Sy, N. H. Lovell, and S. J. Redmond, "Estimating lower body kinematics using a lie group constrained extended Kalman filter and reduced IMU count," *IEEE Sensors J.*, vol. 21, no. 18, pp. 20969–20979, Sep. 2021.
- [136] L. Sy et al., "Estimating lower limb kinematics using a reduced wearable sensor count," *IEEE Trans. Biomed. Eng.*, vol. 68, no. 4, pp. 1293–1304, Apr. 2021.
- [137] A. Atrsaei, H. Salarieh, A. Alasty, and M. Abediny, "Human arm motion tracking by inertial/magnetic sensors using unscented Kalman filter and relative motion constraint," *J. Intell. Robot. Syst.*, vol. 90, nos. 1–2, pp. 161–170, May 2018.
- [138] S. Šlajpah, R. Kamnik, and M. Munih, "Human motion kinematics assessment using wearable sensors," in *Advances in Robot Kinematics*, J. Lenarcic and O. Khatib, Ed. Cham, Switzerland: Springer, 2014, pp. 171–180.
- [139] Z.-B. Wang, L. Yang, Z.-P. Huang, J.-K. Wu, Z.-Q. Zhang, and L.-X. Sun, "Human motion tracking based on complementary Kalman filter," in *Proc. IEEE 14th Int. Conf. Wearable Implant. Body Sensor Netw. (BSN)*, May 2017, pp. 55–58.

- [140] H. T. Butt, M. Pancholi, M. Musahl, P. Murthy, M. A. Sanchez, and D. Stricker, "Inertial motion capture using adaptive sensor fusion and joint angle drift correction," in *Proc. 22th Int. Conf. Inf. Fusion* (*FUSION*), Jul. 2019, pp. 1–8.
- [141] C. Schiefer et al., "Optimization of inertial sensor-based motion capturing for magnetically distorted field applications," J. Biomech. Eng., vol. 136, no. 12, Dec. 2014, Art. no. 121008.
- [142] A. Saito, S. Kizawa, Y. Kobayashi, and K. Miyawaki, "Pose estimation by extended Kalman filter using noise covariance matrices based on sensor output," *ROBOMECH J.*, vol. 7, no. 1, pp. 1–11, Dec. 2020.
- [143] H. Yang, S. Liu, T. Luo, H. Liang, J. Zou, and L. Zhao, "Research on human motion monitoring method based on multi-joint constraint filter model," *IEEE Sensors J.*, vol. 21, no. 9, pp. 10989–10999, May 2021.
- [144] H. Fourati, N. Manamanni, L. Afilal, and Y. Handrich, *Rigid Body Motion Capturing by Means of a Wearable Inertial and Magnetic MEMS Sensor Assembly—From Reconstitution of the Posture Toward Dead Reckoning: An Application in Bio-Logging.* Boca Raton, FL, USA: CRC Press, 2017.
- [145] P. N. Pathirana, M. S. Karunarathne, G. L. Williams, P. T. Nam, and H. Durrant-Whyte, "Robust and accurate capture of human joint pose using an inertial sensor," *IEEE J. Transl. Eng. Health Med.*, vol. 6, pp. 1–11, 2018.
- [146] N. Sim, C. Gavriel, W. W. Abbott, and A. A. Faisal, "The head mouse—Head gaze estimation 'in-the-wild' with low-cost inertial sensors for BMI use," in *Proc. Int. IEEE/EMBS Conf. Neural Eng.* (NER), Nov. 2013, pp. 735–738.
- [147] F. J. Wouda, M. Giuberti, N. Rudigkeit, B.-J.-F. van Beijnum, M. Poel, and P. H. Veltink, "Time coherent full-body poses estimated using only five inertial sensors: Deep versus shallow learning," *Sensors*, vol. 19, no. 17, p. 3716, Aug. 2019.
- [148] D. Nagaraj, R. Dobinson, and D. Werth, "Towards kinematically constrained real time human pose estimation using sparse IMUs," in *Proc. CEUR Workshop*, vol. 2846, 2021, pp. 1–4.
- [149] Z.-Q. Zhang, W.-C. Wong, and J.-K. Wu, "Ubiquitous human upperlimb motion estimation using wearable sensors," *IEEE Trans. Inf. Technol. Biomed.*, vol. 15, no. 4, pp. 513–521, Jul. 2011.
- [150] D. Álvarez, J. C. Alvarez, R. C. González, and A. M. López, "Upper limb joint angle measurement in occupational health," *Comput. Methods Biomech. Biomed. Eng.*, vol. 19, no. 2, pp. 159–170, Jan. 2016.
- [151] K. Liu, T. Liu, K. Shibata, and Y. Inoue, "Visual estimation of lower limb motion using physical and virtual sensors," in *Proc. IEEE Int. Conf. Inf. Autom.*, Jun. 2010, pp. 179–184.
- [152] J. Li et al., "Real-time human motion capture based on wearable inertial sensor networks," *IEEE Internet Things J.*, vol. 9, no. 11, pp. 8953–8966, Jun. 2022.
- [153] M. Zabat, N. Ouadahi, A. Youyou, A. Ababou, and N. Ababou, "Digital inclinometer for joint angles measurements with a real-time 3D-animation," in *Proc. IEEE 12th Int. Multi-Conf. Syst., Signals Devices (SSD)*, Mar. 2015, pp. 1–6.
- [154] K. Liu, I. Yoshio, and S. Kyoko, "Visual and quantitative analysis of lower limb 3D gait posture using accelerometers and magnetometers," in *Proc. IEEE Int. Conf. Mechatronics Autom. (ICMA)*, Aug. 2010, pp. 1420–1425.
- [155] H. T. Butt, B. Taetz, M. Musahl, M. A. Sanchez, P. Murthy, and D. Stricker, "Magnetometer robust deep human pose regression with uncertainty prediction using sparse body worn magnetic inertial measurement units," *IEEE Access*, vol. 9, pp. 36657–36673, 2021.
- [156] D. Simon, Optimal State Estimation: Kalman, H Infinity, and Nonlinear Approaches. Hoboken, NJ, USA: Wiley, 2006.
- [157] R. G. Brown, "Integrated navigation systems and Kalman filtering: A perspective," *Navigation*, vol. 19, no. 4, pp. 355–362, Dec. 1972.
- [158] C. E. Rasmussen, "Gaussian processes in machine learning," in *Proc. Summer School Mach. Learn.* Berlin, Germany: Springer, 2003, pp. 63–71.
- [159] L. Rokach and O. Maimon, "Decision trees," in *Data Mining and Knowledge Discovery Handbook*. New York, NY, USA: Springer, 2005, pp. 165–192.
- [160] B. Schölkopf et al., Learning With Kernels: Support Vector Machines, Regularization, Optimization, and Beyond. Cambridge, MA, USA: MIT Press, 2002.
- [161] B. Schölkopf, A. J. Smola, R. C. Williamson, and P. L. Bartlett, "New support vector algorithms," *Neural Comput.*, vol. 12, no. 5, pp. 1207–1245, May 2000.

- [162] A. K. Jain, J. Mao, and K. M. Mohiuddin, "Artificial neural networks: A tutorial," *Computer*, vol. 29, no. 3, pp. 31–44, Mar. 1996.
- [163] M. F. Alghifari, T. S. Gunawan, and M. Kartiwi, "Speech emotion recognition using deep feedforward neural network," *Indonesian J. Electr. Eng. Comput. Sci.*, vol. 10, no. 2, pp. 554–561, 2018.
- [164] A. V. Oppenheim et al., Signals & Systems. London, U.K.: Pearson, 1997.
- [165] K. Simonyan and A. Zisserman, "Very deep convolutional networks for large-scale image recognition," 2014, arXiv:1409.1556.
- [166] K. He, X. Zhang, S. Ren, and J. Sun, "Deep residual learning for image recognition," in *Proc. IEEE Conf. Comput. Vis. Pattern Recognit.* (CVPR), Jun. 2016, pp. 770–778.
- [167] B. A. Pearlmutter, "Learning state space trajectories in recurrent neural networks," *Neural Comput.*, vol. 1, no. 2, pp. 263–269, Jun. 1989.
- [168] S. Hochreiter and J. Schmidhuber, "Long short-term memory," Neural Comput., vol. 9, no. 8, pp. 1735–1780, 1997.
- [169] K. Cho et al., "Learning phrase representations using RNN encoderdecoder for statistical machine translation," 2014, arXiv:1406.1078.
- [170] T. Lin, B. G. Horne, P. Tino, and C. L. Giles, "Learning long-term dependencies in NARX recurrent neural networks," *IEEE Trans. Neural Netw.*, vol. 7, no. 6, pp. 1329–1338, Nov. 1996.
- [171] (2020). OpenSim. Accessed: May 23, 2020. [Online]. Available: https://opensim.stanford.edu/
- [172] J. J. Uicker, J. Denavit, and R. S. Hartenberg, "An iterative method for the displacement analysis of spatial mechanisms," *J. Appl. Mech.*, vol. 31, no. 2, pp. 309–314, Jun. 1964.
- [173] J. B. Lee, B. Burkett, R. B. Mellifont, and D. A. James, "The use of micro-electro-mechanical-systems technology to assess gait characteristics," in *The Impact of Technology on Sport II*. London, U.K.: Taylor & Francis, 2008, pp. 181–186.
- [174] L. W. F. Sy, N. H. Lovell, and S. J. Redmond, "Estimating lower limb kinematics using a lie group constrained extended Kalman filter with a reduced wearable IMU count and distance measurements," *Sensors*, vol. 20, no. 23, pp. 1–28, 2020.
- [175] Vicon. (2020). Vicon Motion Capture. Accessed: Jan. 28, 2020. [Online]. Available: https://www.vicon.com/
- [176] OptiTrack. (2020). Optitrack—Motive—Optical Motion Capture Software. Accessed: Jun. 5, 2020. [Online]. Available: https://optitrack.com/software/motive/
- [177] (2023). Xsens—Movella. Accessed: Mar. 23, 2023. [Online]. Available: https://www.movella.com/
- [178] M. Crabolu, D. Pani, L. Raffo, M. Conti, P. Crivelli, and A. Cereatti, "In vivo estimation of the shoulder joint center of rotation using magneto-inertial sensors: MRI-based accuracy and repeatability assessment," *Biomed. Eng. OnLine*, vol. 16, no. 1, pp. 1–18, Dec. 2017.
- [179] S. G. de Villa, A. J. Martín, and J. J. G. Domínguez, "Implementation of a lower-limb model for monitoring exercises in rehabilitation," in *Proc. IEEE Int. Symp. Med. Meas. Appl. (MeMeA)*, Jun. 2019, pp. 1–6.
- [180] E. Frick and S. Rahmatalla, "Joint center estimation using single-frame optimization: Part 1: Numerical simulation," *Sensors*, vol. 18, no. 4, pp. 1–17, 2018.
- [181] E. Frick and S. Rahmatalla, "Joint center estimation using singleframe optimization: Part 2: Experimentation," *Sensors*, vol. 18, no. 8, pp. 1–22, 2018.
- [182] S. García-de-Villa, A. Jiménez-Martín, and J. J. García-Domínguez, "Novel IMU-based adaptive estimator of the center of rotation of joints for movement analysis," *IEEE Trans. Instrum. Meas.*, vol. 70, pp. 1–11, 2021.
- [183] S. G. de Villa, A. J. Martín, and J. J. G. Domínguez, "Adaptive IMUbased calibration of the center of joints for movement analysis: One case study," in *Proc. IEEE Int. Symp. Med. Meas. Appl. (MeMeA)*, Jun. 2020, pp. 1–10.
- [184] C. J. Lee and J. K. Lee, "Wearable IMMU-based relative position estimation between body segments via time-varying segment-to-joint vectors," *Sensors*, vol. 22, no. 6, p. 2149, Mar. 2022.
- [185] S. García-de-Villa, A. Jiménez-Martín, and J. J. García-Domínguez, "A database of physical therapy exercises with variability of execution collected by wearable sensors," *Sci. Data*, vol. 9, no. 1, pp. 1–13, Jun. 2022.
- [186] G. Santos, M. Wanderley, T. Tavares, and A. Rocha, "A multisensor human gait dataset captured through an optical system and inertial measurement units," *Sci. Data*, vol. 9, Sep. 2022, Art. no. 545.
- [187] T. Liu, Y. Inoue, and K. Shibata, "Development of a wearable sensor system for quantitative gait analysis," *Measurement*, vol. 42, no. 7, pp. 978–988, Aug. 2009.



Sara García-de-Villa received the B.E. degree in biomedical engineering from the Polytechnic University of Madrid, Madrid, Spain, in 2016, the M.Sc. degree in biomedical engineering with intensification in biomechanics from the Polytechnic University of Valencia, Valencia, Spain, in 2017, and the Ph.D. degree in electronic control systems from the University of Alcalá, Alcalá de Henares, Madrid, in 2022.

She is currently an Assistant Professor with the Department of Signal Processing and Communica-

tions, Universidad Rey Juan Carlos, Madrid. Her current research areas focus on human motion analysis, applied to the monitoring of physical therapies.



Ana Jiménez-Martín (Member, IEEE) received the degree in physics from the Complutense University of Madrid, Madrid, Spain, in 1994, and the Materials Engineering and Ph.D. degrees from the Technical University of Madrid, Madrid, in 1998 and 2003, respectively.

Since 2004, she has been an Associate Professor with the Department of Electronics, University of Alcalá, Alcalá de Henares, Spain. During her university activity, she has participated in numerous research projects covering very diverse areas, such

as microelectronics (growth and characterization of GaN-related devices), ultrasonic sensors, multisensorial systems, and indoor and underwater local positioning systems. Her current research work focuses on ambient intelligence for independent living and healthy aging.



**David Casillas-Pérez** received the B.S. degree in telecommunication engineering, the M.S. and Ph.D. degrees in electronic control systems from the Universidad de Alcalá, Alcalá de Henares, Spain, in 2013, 2014, and 2019, respectively.

He is currently an Assistant Professor with the Department of Signal Processing and Communications, Universidad Rey Juan Carlos, Madrid, Spain. His research is focused on the development of machine learning algorithms with applications in different fields, such as computer vision, mobile

communication systems, or renewable energy systems.



Juan Jesús García-Domínguez (Senior Member, IEEE) received the degree in telecommunications engineering from the Polytechnic University of Valencia, Valencia, Spain, and the European Ph.D. degree in telecommunications engineering from the University of Alcalá, Alcalá de Henares, Spain, in 1999.

He is currently a Professor with the Department of Electronics, University of Alcalá. His research interests cover the areas of multisensorial systems and positioning systems. Currently, he is focusing

on ambient intelligence for independent living, applied to the evaluation of frailty in the elderly and the early detection of the onset of physical and cognitive impairment.

Dr. García-Domínguez was the Awarded Best Doctoral Thesis in 2006.